

Asymptotics for metamaterial cavities and their effect on scattering

Camille Carvalho, Zoïs Moitier

▶ To cite this version:

Camille Carvalho, Zoïs Moitier. Asymptotics for metamaterial cavities and their effect on scattering. 2020. hal-02965993v1

HAL Id: hal-02965993 https://hal.science/hal-02965993v1

Preprint submitted on 13 Oct 2020 (v1), last revised 6 Dec 2022 (v3)

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

ASYMPTOTICS FOR METAMATERIAL CAVITIES AND THEIR EFFECT ON SCATTERING

CAMILLE CARVALHO AND ZOS MOITIER

ABSTRACT. It is well-known that optical cavities can exhibit localized phenomena responsible for numerical instabilities in approximating scattering problems. In classic optical cavities, those localized phenomena concentrate at the inner boundary of the cavity and are called whispering gallery modes. In this paper we consider cavities made of Negative-Index Metamaterials (NIM). Those manufactured materials can exhibit unusual optical properties, leading to the appearance of exotic waves at the interface of metamaterial-dielectric interface, such as surface plasmons. There is a great interest in controlling and guiding surface plasmons to design the next-generation of nano-antennas, sensors, and others. Surface plasmons have been usually characterized in the context of the specific quasi-static case. Using asymptotic analysis, we provide a systematic mathematical characterization of emerging surface plasmons for general metamaterial cavities for the full wave problem, and answer the following questions: can surface plasmons be always excited in scattering problems? Can they create numerical instabilities? Our asymptotic analysis reveals that the nature of surface plasmons changes depending on the metamaterial's properties, leading to different effects on scattering problems. Numerical results for several metamaterial cavities are provided.

CONTENTS

1.	Introducti	on	1
2.	Mathematical problem and preliminaries		2
3.	Case of a circular cavity		4
4.	4. Asymptotic for metamaterial cavities		10
5.	. Back to the scattering problem		19
6.	6. Conclusions		21
Appendix A. Well-posedness of the scattering problem		22	
Appendix B. Proofs and additional results for the asymptotic expansions		22	
Acknowledgments			24
References			25

1. INTRODUCTION

Optical micro-resonators, commonly involving dielectric cavities, play an important role in many applications in photonics and others [27]. Optical cavities have been extensively studied over the past decades as they offer the opportunity to confine light in small volumes. This is possible in particular when the resonator supports the so-called Whispering Gallery Modes (WGM) [37]. With the development of metamaterials, such as the Negative-Index Metamaterials (NIM) which exhibit unusual optical

Date: October 13, 2020.

²⁰¹⁰ Mathematics Subject Classification. 78M35; 35P25; 35C20; 78M10.

Key words and phrases. Helmholtz Equation; Resonances; Scattering; Surface Plasmon.

properties (for instance a negative effective permittivity ε and/or a negative effective permeability μ), there is a great interest in modeling metamaterial cavities to confine and control light. In particular, at optical frequencies, localized surface waves called surface plasmons can arise at dielectric-metamaterial interfaces (as well for some dielectric-metal interfaces) [31]. The field of plasmonics is very active as surface plasmons offer strong light enhancement, with applications to next-generation sensors, antennas, high-resolution imaging, cloaking and other [38]. However surface plasmons are very sensitive to the geometry and therefore challenging to capture, experimentally and numerically [8, 28].

Mathematically, surface plasmons are solutions of the homogeneous Maxwell's equations, they are oscillatory waves along the dielectric-metamaterial interface while exponentially decreasing in the transverse direction. For simple interface geometries their expression is known explicitly, however in general surface plasmons have been mainly characterized and investigated in the context of the quasi-static approximation (e.g. [12, 25, 8, 4, 16, 14, 13]). While the quasi-static case offers valuable insights into their behavior, the connection with exciting the characterized surface plasmons in the context of light scattering by metamaterials is less clear.

It is well-known that the approximation of light scattering in dielectric optical microcavities can be drastically affected by WGM, in particular if the excitation wavenumber of the source is close to a WGM resonance [32, 6].

In the spirit of [6], we carry out in this paper an asymptotic analysis to systematically characterize surface plasmons arising in general two-dimensional metamaterial cavities (arbitrary smooth shape, arbitrary permittivity function) for the full wave problem (time-harmonic Maxwell's equations in Transverse Magnetic polarization), and we answer the following questions: Can surface plasmons be always excited in scattering problems? Can we numerically observe surface plasmons in practice? Can they create numerical instabilities? Our asymptotic analysis reveals that the spectral nature of surface plasmons changes depending on the metamaterial's properties, leading to different effects on scattering problems. Additionally, this analysis provides guidance about when surface plasmons can be excited in practice.

The paper is organized as follows. We present the considered scattering problem and known results in Section 2. Section 3 presents how to characterize surface plasmons and their effect on scattering problems for the model case of a circular metamaterial cavity. Section 4 provides the general asymptotic analysis of emerging surface plasmons. Section 5 presents numerical results based on the asymptotic analysis, and Section 6 presents our concluding remarks. Appendix A provides theoretical results regarding well-posedness of the scattering problem and Appendix B provides additional results and proofs needed in Section 4.

2. MATHEMATICAL PROBLEM AND PRELIMINARIES

2.1. Scattering by a metamaterial cavity. We consider a cavity $\Omega \subset \mathbb{R}^2$, an open bounded connected set with smooth boundary Γ . The optical cavity Ω is characterized by a permeability $\mu = 1$, and a permittivity $\varepsilon := \varepsilon_c < 0$ with $\varepsilon_c \in \mathscr{C}^{\infty}(\overline{\Omega})$. The cavity is embedded in an homogeneous background characterized by $\mu = 1$, and $\varepsilon = 1$. We consider the scattering problem of a plane wave by the cavity Ω . Let $u^{\text{in}} : x \mapsto e^{ik d \cdot x}$, the plane wave with wavenumber k > 0 and direction $d \in \mathbb{S}^1$, be the incident wave. The scattering problem writes: Find $u \in \mathrm{H}^{1}_{\mathrm{loc}}(\mathbb{R}^{2})$ such that $u = u^{\mathrm{in}} + u^{\mathrm{sc}}$ and

$$\begin{cases} -\operatorname{div}\left(\varepsilon^{-1}\nabla u\right) - k^{2}u = 0 & \text{in } \mathbb{R}^{2} \\ \left[u\right]_{\Gamma} = 0 & \text{and} \quad \left[\varepsilon^{-1}\partial_{n}u\right]_{\Gamma} = 0 & \text{across } \Gamma \\ u^{\mathsf{sc}} & k\text{-outgoing} \end{cases}$$
(2.1)

with $\mathrm{H}^{1}_{\mathrm{loc}}(\mathbb{R}^{2}) \coloneqq \{ u \in \mathrm{L}^{2}_{\mathrm{loc}}(\mathbb{R}^{2}) \mid \forall \chi \in \mathscr{C}^{\infty}_{\mathrm{comp}}(\mathbb{R}^{2}), \ \chi u \in \mathrm{H}^{1}(\mathbb{R}^{2}) \}, n : \Gamma \to \mathbb{S}^{1} \text{ is the unit}$ normal vector outward to Ω , and the function $\varepsilon \in \mathrm{L}^{\infty}(\mathbb{R}^{2})$ is given by

$$\varepsilon(x) = \begin{cases} \varepsilon_{\mathsf{c}}(x) < 0 & \text{if } x \in \overline{\Omega} \\ 1 & \text{if } x \in \mathbb{R}^2 \setminus \overline{\Omega} \end{cases}.$$
(2.2)

Given X, we denote $[X]_{\Gamma}(\gamma) = \lim_{x \to \gamma^+} X(x) - \lim_{x \to \gamma^-} X(x)$, for any $\gamma \in \Gamma$, the jump condition across Γ . The jump conditions $[u]_{\Gamma} = 0$ and $[\varepsilon^{-1} \partial_n u]_{\Gamma} = 0$ will be referred to as the *transmission conditions*. We say that v is k-outgoing if it satisfies the outgoing wave condition:

$$v(r,\theta) = \sum_{m \in \mathbb{Z}} w_m(r) \,\mathsf{e}^{\mathsf{i}m\theta} = \sum_{m \in \mathbb{Z}} c_m \,\mathsf{H}_m^{(1)}(kr) \,\mathsf{e}^{\mathsf{i}m\theta} \tag{2.3}$$

with polar coordinates (r, θ) such that $r > \sup_{x \in \Omega} |x|, \theta \in \mathbb{R}/2\pi\mathbb{Z}, H_m^{(1)}$ the Hankel function of the first kind of order m, and $(c_m)_{m \in \mathbb{Z}} \in \mathbb{C}^{\mathbb{Z}}$.

Remark 2.1. The condition v is k-outgoing defined in Eq. (2.3) is equivalent to v satisfying the so-called Sommerfeld radiitation condition, if and only if k > 0. This outgoing condition is more general, and will be also used for the associated spectral problem, where one can have $k \in \mathbb{C}$.

In classic scattering problem (i.e. when $\varepsilon(x) > 0$, $\forall x \in \mathbb{R}^2$), it is well known that Problem Eq. (2.1) is well-posed [15, 21]. In our case ε defined as Eq. (2.2) is *sign-changing*, and the problem may be ill-posed. Using the T-coercivity theory [10, 8, 9], one can establish the following Lemma (see Appendix A for details):

Lemma 2.2. Problem (2.1) is well-posed if and only if $\varepsilon_{\mathsf{c}}|_{\Gamma} \neq -1$. Moreover there exists a constant C(k) > 0 such that

$$\|u^{\mathsf{sc}}\|_{L^{2}(\mathbb{D}(0,\rho))} \le C(k) \|u^{\mathsf{in}}\|_{L^{2}(\mathbb{D}(0,\rho))}$$
 (2.4)

where $\mathbb{D}(0,\rho) \supseteq \Omega$ is the disk of center 0 and radius ρ .

2.2. Resonances of a dielectric-metamaterial open cavity. As done for classical cavities, it is essential to study the spectral problem associated to Problem (2.1) to identify if resonances appear and study their affect if one chooses k "close" to a resonance [41, 22, 32]. We define the operator $P := -\operatorname{div}(\varepsilon^{-1}\nabla)$ on $L^2(\mathbb{R}^2)$ with domain $D(P) := \{u \in L^2(\mathbb{R}^2) \mid \operatorname{div}(\varepsilon^{-1}\nabla u) \in L^2(\mathbb{R}^2)\}$. The spectral problem associated to (2.1) writes: Find $\ell \in \mathbb{C}^{\frac{1}{2}}$ and $u \in D_{\operatorname{loc}}(P), u \not\equiv 0$ such that

$$\begin{cases} Pu = \ell^2 u & \text{in } \mathbb{R}^2\\ [u]_{\Gamma} = 0 \quad \text{and} \quad [\varepsilon^{-1} \partial_{\nu} u]_{\Gamma} = 0 \quad \text{across } \Gamma ,\\ u \quad \ell \text{-outgoing} \end{cases}$$
(2.5)

where $\mathbb{C}^{\frac{1}{2}} := \{z \in \mathbb{C} \mid \arg(z) \in \left(-\frac{\pi}{2}, \frac{\pi}{2}\right]\}$. We set the branch cut to be \mathbb{R}_{-} , from now on complex roots are uniquely defined. In order to capture eigenvalues and resonances, we study Problem (2.5) in an extended framework $D_{\text{loc}}(P) \supseteq D(P)$, with

$$D_{\rm loc}(P) \coloneqq \left\{ u \in \mathcal{L}^2_{\rm loc}(\mathbb{R}^2) \mid \forall \chi \in \mathscr{C}^\infty_{\rm comp}(\mathbb{R}^2), \ \chi u \in D(P) \right\}$$

Remark 2.3. With the chosen branch cut, the outgoing condition gives that u is a solution of Eq. (2.5) in $L^2(\mathbb{R}^2)$ if, and only if, $\ell \in \mathbb{C}^{i+}_* := \{z \in \mathbb{C}^{\frac{1}{2}} \mid \Im(z) > 0\}$. This is coming from the fact that, for $m \in \mathbb{Z}$, the asymptotic behavior of $H_m^{(1)}$ as $r \to +\infty$ is given by $\left| H_m^{(1)}(\ell r) \right| \sim \sqrt{\frac{2}{\pi |\ell| r}} e^{-\Im(\ell) r}$. We also define $\mathbb{C}^{i-} := \{z \in \mathbb{C}^{\frac{1}{2}} \mid \Im(z) \leq 0\}$. As a consequence we have $u \in D(P)$ when associated to $\ell \in \mathbb{C}^{i+}_*$, and $u \in D_{\text{loc}}(P)$ when associated to $\ell \in \mathbb{C}^{i-}_*$.

Using T-coercivity arguments [18, 16], one can show that, if $\varepsilon_{\mathsf{c}}|_{\Gamma} \neq -1$, then the operator (P, D(P)) is self-adjoint on $L^2(\mathbb{R}^2)$ and its essential spectrum is \mathbb{R}_+ (we refer to [17] for details). However, contrary to the case with classical cavities [6], P doesn't admit a lower bound, it can exhibit negative eigenvalues, as well as *interface modes* related to surface plasmons [16]. The rest of the paper is dedicated to: (i) investigate what types of resonances holds Problem Eq. (2.5) (explicitly or asymptotically), and (ii) identify if the computations can be deteriorated close to resonances (related to surface plasmons, or others). We start with a simple case where calculations are explicit.

Remark 2.4. In this paper, we consider the Helmholtz equation case, we do not assume the quasi-static approximation. The considered spectral problem, where the spectral parameter is the wavenumber ℓ , differs from the quasi-static spectral problem, where the spectral parameter is the permittivity ε_{c} [25, 13].

3. Case of a circular cavity

We consider $\Omega = \mathbb{D}(0, R)$ the disk of center 0 and radius R > 0. We set $\varepsilon \equiv 1$ outside the cavity, and for simplicity we fix $\varepsilon = \varepsilon_{c} \equiv -\eta^{2}$ inside, with constant $\eta > 0$ (in other words $\eta \coloneqq \sqrt{-\varepsilon_{c}}$).

3.1. Scattering by a disk. We take $d = (0, 1)^{\intercal}$. Note that we can get the solution for a direction d' by doing a rotation that transforms d into d'. In the following, J_m denotes the Bessel function of the first kind of order m, and I_m denotes the modified Bessel function of the first kind of order m. Taking advantage of the geometry, we look for solution of Problem Eq. (2.1) of the form:

$$u(x) = u(r,\theta) = \sum_{m \in \mathbb{Z}} u_m(r,\theta) = \sum_{m \in \mathbb{Z}} w_m(r) \,\mathsf{e}^{\mathsf{i}m\theta},\tag{3.1}$$

with $(r, \theta) \in \mathbb{R}_+ \times \mathbb{R}/2\pi\mathbb{Z}$ the polar coordinates corresponding to the Cartesian coordinates x, and $w_m(r) = \frac{1}{2\pi} \int_0^{2\pi} u(r, \theta) e^{-im\theta} d\theta$, $m \in \mathbb{Z}$. Similarly we write $u^{in}(x) = \sum_{m \in \mathbb{Z}} w_m^{in} e^{im\theta}$ and $u^{sc}(x) = \sum_{m \in \mathbb{Z}} w_m^{sc}(r) e^{im\theta}$.

Remark 3.1. The Jacobi-Anger expansion [35, Eq. 10.12.1] states that

$$u^{\text{in}}(x) = e^{ik \, d \cdot x} = \sum_{m \in \mathbb{Z}} \mathsf{J}_m(kr) \, e^{im\left(\theta - \phi + \frac{\pi}{2}\right)}, \qquad \text{with } d = (\cos(\phi), \sin(\phi))^{\intercal}. \tag{3.2}$$

The series in (3.2) converges absolutely on every compact set of \mathbb{R}^2 . With $d = (0, 1)^{\intercal}$, we obtain $w_m^{\text{in}}(r) = \mathsf{J}_m(kr), \forall m \in \mathbb{Z}$.

Plugging in Problem Eq. (2.1), we obtain a family of 1D problems indexed by $m \in \mathbb{Z}$: Find $w_m \in \mathrm{H}^1_{\mathrm{loc}}(\mathbb{R}_+, r \, \mathrm{d}r)$ such that

$$\begin{cases} -\frac{1}{r}\partial_r \left(r \,\partial_r w_m\right) + \frac{m^2}{r^2} w_m - \varepsilon \,k^2 \,w_m = 0 & \text{in } (0, R) \cup (R, +\infty) \\ [w_m]_{\{R\}} = 0 & \text{and} & \left[\varepsilon^{-1} \,w'_m\right]_{\{R\}} = 0 & \text{across } \{R\} \\ w'_0(0) = 0 & \text{and} & w^{\text{sc}}_m(r) \propto \mathsf{H}_m^{(1)}(kr) & r > R \end{cases}$$
(3.3)

with \propto meaning "up to a constant". For $m \neq 0$, the term $\frac{m^2}{r^2} w_m$ imposes a homogeneous Dirichlet boundary condition at zero [7]. Since the solution is continuous at r = 0, using the outgoing wave condition and the transmission conditions at r = R, we write

$$w_m^{\mathsf{sc}}(r) = \begin{cases} \alpha_m \operatorname{I}_m(\eta \, k \, r) - \operatorname{J}_m(k \, r) & \text{if } r \le R\\ \beta_m \operatorname{H}_m^{(1)}(k \, r) & \text{if } r > R \end{cases}$$
(3.4)

with (α_m, β_m) solution of the system $M_m(\eta, k R)(\alpha_m, \beta_m)^{\mathsf{T}} = (\mathsf{J}_m(k R), \mathsf{J}'_m(k R))^{\mathsf{T}}$ where

$$M_m(\eta, k R) = \begin{pmatrix} \mathsf{I}_m(\eta \, k \, R) & -\mathsf{H}_m^{(1)}(k \, R) \\ -\eta^{-1} \, \mathsf{I}'_m(\eta \, k \, R) & -\mathsf{H}_m^{(1)'}(k \, R) \end{pmatrix}.$$
 (3.5)

Remark 3.2. Since k > 0 and the problem is well-posed for $\eta \neq 1$, coefficients (α_m, β_m) are uniquely defined and det $(M_m(\eta, k R)) \neq 0$, with

$$\det(M_m(\eta, z)) \coloneqq -\eta^{-1} \,\mathsf{I}'_m(\eta \, z) \,\mathsf{H}^{(1)}_m(z) - \mathsf{I}_m(\eta \, z) \mathsf{H}^{(1)'}_m(z), \quad \forall z \in \mathbb{C}^*.$$
(3.6)

Now that we have an explicit expression of u^{sc} , we can analyze its behavior for various wavenumbers k and permittivities ε_c . For numerical purposes, we define u_k the approximate solution of Problem Eq. (2.1) associated to the wavenumber k, of order M: $u_k = \sum_{m=-M}^{M} w_m(r) e^{im\theta}$, and we define the sequence $(u_k)_k := (u_k^{in})_k + (u_k^{sc})_k$. We choose here M = 32, so that $(u_k)_k$ converges to order 10^{-16} . We define $L^2_{\rho} := L^2(\mathbb{D}(0, \rho))$ for $\rho > R$ and the function $N_{\varepsilon,\rho}$ defined by

$$N_{\varepsilon,\rho}(k) = \frac{\|u_k^{\mathsf{sc}}\|_{\mathrm{L}^2_{\rho}}}{\|u_k^{\mathsf{in}}\|_{\mathrm{L}^2_{\rho}}}$$

The function $k \mapsto N_{\varepsilon,\rho}(k)$ characterizes the well-posedness constant as explained in Remark A.2. We consider the scattering of a plane wave by the unit disk (R = 1) and choose $\rho = 2$. Figure 1 represents the log plot of $N_{\varepsilon,2}$ with respect to k, for various values of ε_{c} . One observes that

- For $-1 < \varepsilon_{c} < 0$, $N_{\varepsilon,2}$ remains bounded.
- For $\varepsilon_{\mathbf{c}} < -1$, there exists a sequence k_m such that $N_{\varepsilon,2}(k_m)$ peaks and the sequence $(N_{\varepsilon,2}(k_m))_{m\geq 1}$ grows exponentially.

To understand the observed instabilities when $\varepsilon_{c} < -1$, let us take a closer look at u_{k}^{sc} for $k = k_{m}$ and $k \neq k_{m}$, for a given m. Figure 2 represents the modulus of u_{k}^{sc} for $k = \{k_{12} - 0.01, k_{12}, k_{12} + 0.01\}$ and $\varepsilon_{c} \equiv -1.1$. One observes for $k = k_{m}$ an intensely localized interface behavior for the scattered field with roughly four orders of magnitude compared to the amplitude's field when $k \neq k_{m}$. Moreover the localized behavior is oscillatory along the interface contrary to when $k \neq k_{m}$. This localized behavior indicates the appearance of a surface plasmon. The above results provide the following:

• Even if the scattering problem is well-posed for all k > 0, we can have strong instabilities for a sequence of wavenumbers $(k_m)_m$.



FIGURE 1. Log plot of $k \mapsto N_{\varepsilon,2}(k)$ with respect to k for $\varepsilon_c \in \{-0.9, -0.8, -0.7\}$ (left), for $\varepsilon_c \in \{-1.3, -1.2, -1.1\}$ (right). The value k_{12} marked on the graphs corresponds to the reference value used in Figure 2.



FIGURE 2. Modulus of the scattered field u^{sc} for a disk of radius 1 with $\varepsilon_{\text{c}} = -1.1$ ($\eta = \sqrt{1.1}$), and $k = k_{12} - 0.01$ (left column), k_{12} (middle), and $k_{12} + 0.01$ (right column).

• Those instabilities arise only for $\varepsilon_{c} < -1$, and are due to surface plasmons.

Can we characterize the sequence $(k_m)_{m\geq 1}$? Can we justify that the instabilities are caused by surface plasmons? In what follows we investigate the associated spectral problem and use semi-classical analysis to answer those questions.

3.2. Eigenvalues and resonances for the disk. Proceeding similarly as in Section 3.1, the spectral problem Eq. (2.5) set on a disk can be rewritten as a family of 1D problems indexed by $m \in \mathbb{Z}$: Find $(\ell, w_m) \in \mathbb{C}^{\frac{1}{2}} \times \mathrm{H}^{1}_{\mathrm{loc}}(\mathbb{R}_+, r \, \mathrm{d}r) \setminus \{0\}$, such that

$$\begin{cases} -\frac{1}{r}\partial_r \left(r \,\partial_r w_m\right) + \frac{m^2}{r^2} w_m = \varepsilon \ell^2 w_m & \text{in } (0, R) \cup (R, +\infty) \\ [w_m]_{\{R\}} = 0 & \text{and} & \left[\varepsilon^{-1} w'_m\right]_{\{R\}} = 0 & \text{across } \{R\} \\ w'_0(0) = 0 & \text{and} & w_m(r) \propto \mathsf{H}_m^{(1)} \left(\ell r\right) & r > R \end{cases}$$
(3.7)

Similarly, we write

$$w_m(r) = \begin{cases} \alpha_m \operatorname{I}_m(\eta \,\ell \, r) & \text{if } r \le R\\ \beta_m \operatorname{H}_m^{(1)}(\ell \, r) & \text{if } r > R \end{cases}$$
(3.8)

however this time, the pair (ℓ, w_m) is solution of Eq. (3.7) if, and only if, there exists $(\alpha_m, \beta_m)^{\intercal} \in \ker(M(\eta, \ell R)) \setminus (0, 0)^{\intercal}$, with $M(\eta, \ell R)$ defined in Eq. (3.5). The values $\ell^2 \in \mathbb{C}$ such that (ℓ, w_m) is a solution of Eq. (3.7) are called resonances, and w_m is called the associated resonant mode. To distinguish eigenvalues from resonances, we will use the fact that a resonant mode associated to an eigenvalue is in $\mathrm{H}^1(\mathbb{R}_+, r \, \mathrm{d}r)$. Given $m \in \mathbb{Z}$, and using Eq. (3.6), we define the set of resonances

$$\mathcal{R}[\varepsilon_{\mathsf{c}}, R](m) = \left\{ \ell^2 \in \mathbb{C} \mid \det(M_m(\eta, \ell R) = 0 \text{ and } \ell \in \mathbb{C}^{\frac{1}{2}} \right\}.$$
(3.9)

Finally we define the set of resonances of Problem Eq. (2.5)

$$\mathcal{R}[\varepsilon_{\mathsf{c}}, R] \coloneqq \bigcup_{m \in \mathbb{Z}} \mathcal{R}[\varepsilon_{\mathsf{c}}, R](m).$$
(3.10)

Remark 3.3. Given $\ell^2 \in \mathcal{R}[\varepsilon_{\mathsf{c}}, R](m)$, one finds $\alpha_m = c$ and $\beta_m = c \frac{\mathsf{I}_m(\eta \,\ell \, R)}{\mathsf{H}_m^{(1)}(\ell \, R)}$ with $c \in \mathbb{C}^*$ since the resonant modes are defined up to some normalization.

Remark 3.4. Since $I_{-m} = I_m$ and $H_{-m}^{(1)} = (-1)^m H_m^{(1)}$, $\forall m \in \mathbb{Z}$, [35, Eq. 10.27.1 and 10.4.2], by symmetry all the resonances ℓ^2 , corresponding to $m \neq 0$, are of multiplicity 2, and the two associated modes are conjugate, given by $u_m(r,\theta) \coloneqq w_m(r) e^{\pm im\theta}$. It turns out $\mathcal{R}[\varepsilon_{\mathsf{c}}, R] \coloneqq \bigcup_{m \in \mathbb{N}} \mathcal{R}[\varepsilon_{\mathsf{c}}, R](m)$.

The resonances sets $(\mathcal{R}[\varepsilon_{c}, R](m))_{m}$ defined in Eq. (3.9) cannot be computed analytically, however one can use contour integration techniques on Eq. (3.6) to compute a subset $\mathcal{R}_{N}[\varepsilon_{c}, R] := \bigcup_{m=0}^{N} \mathcal{R}[\varepsilon_{c}, R](m) \subset \mathcal{R}[\varepsilon_{c}, R]$ (see [30, 36]). Figure 3 represents the set $\mathcal{R}_{64}[\varepsilon_{c}, 1]$ for the unit disk and for various permittivities ε_{c} . The color bar indicates the value of m.



FIGURE 3. Graph of the sets $\mathcal{R}_{64}[-0.9, 1]$ (left) and $\mathcal{R}_{64}[-1.1, 1]$ (right) in the complex plane $(\Re(\ell^2), \Im(\ell^2))$, computed using complex contour integration [36] on the analytic function Eq. (3.6).

In classical cavities, resonances of Problem Eq. (2.5) (at least for $\varepsilon_c > 1$) [6] are split into two categories: inner resonances $\mathcal{R}_{inn}[\varepsilon_c, R]$ associated to resonant modes essentially supported inside the cavity Ω , and outer resonances $\mathcal{R}_{out}[\varepsilon_c, R]$ associated to resonant modes essentially supported in the exterior of the cavity $\mathbb{R}^2 \setminus \overline{\Omega}$. The inner resonance category includes the so-called *Whispering Gallery Modes* (WGM), associated to resonances ℓ_{WGM} such that $-1 \ll \Im(\ell_{WGM}) < 0$ [19]. In particular the approximation of Eq. (2.1) can be deteriorated if one chooses $k = \Re(\ell_{WGM})$, where those modes can be excited [32, Sec. 6.2]. When $\varepsilon_c < 0$ we split the resonances into three categories. From Fig. 3, and using Remark 2.3 we conclude:

- The outer resonances $\mathcal{R}_{out}[\varepsilon_c, 1]$ (represented as triangles in Fig. 3) are resonances with a negative imaginary part.
- The inner resonances $\mathcal{R}_{inn}[\varepsilon_c, 1]$ (represented as dots in Fig. 3) are negative real eigenvalues of the operator P on $L^2(\mathbb{R}^2)$. They contain whispering gallery modes.
- The resonances represented at '+' inside the red rectangles in Fig. 3 are associated to resonant modes essentially supported on the interface Γ (see Fig. 4 for an example). We refer to those modes as surface plasmons, and we call this family the interface resonances $\mathcal{R}_{\mathsf{pla}}[\varepsilon_{\mathsf{c}}, 1]$. We denote the interface resonances $(\ell_m^2)_m$ so that $\mathcal{R}_{\mathsf{pla}}[\varepsilon_{\mathsf{c}}, 1] = \{\ell_m^2 \mid m \in \mathbb{N}^*\}$.



FIGURE 4. Log-Log plots of the radial component $r \mapsto w_m(r)$ of the interface resonant mode $(r, \theta) \mapsto u_m(r, \theta) = w_m(r)e^{im\theta}$ for $\varepsilon_c \in \{-1.1, -0.9\}$ and for m = 6 (c), for m = 12 (f). When $\varepsilon_c < -1$, $w_m(r)$ tends to ∞ as $r \to \infty$, therefore $u_6 \in L^2_{loc}(\mathbb{R}^2)$ ($\ell_6 \in \mathbb{C}^{i-}$ corresponds to a resonance). When $-1 < \varepsilon_c < 0$, $w_6(r)$ tends to 0 as $r \to \infty$, therefore $u_6 \in L^2(\mathbb{R}^2)$ ($\ell_6 \in \mathbb{C}^{i+}$ corresponds to an eigenvalue).

In the end we write $\mathcal{R}[\varepsilon_{c}, R] = \mathcal{R}_{out}[\varepsilon_{c}, R] \cup \mathcal{R}_{inn}[\varepsilon_{c}, R] \cup \mathcal{R}_{pla}[\varepsilon_{c}, R]$. The interface resonances are quite peculiar as their nature changes depending on $\varepsilon_{\rm c}$. As illustrated in Fig. 3, they correspond to negative real eigenvalues when $-1 < \varepsilon_{c} < 0$, while they correspond to complex resonances in \mathbb{C}^{i-} when $\varepsilon_{c} < -1$. One observes that the negative eigenvalues diverge to $-\infty$ as $m \to \infty$, while the real part of the complex resonances diverge toward $+\infty$ as $m \to \infty$, and their negative imaginary part tends to 0 exponentially fast as $m \to \infty$. Additionally, a closer observation gives us that $\Re(\ell_m^2) \propto$ m^2 . Figure 4 represents the behavior of $w_m(r)$ far from the boundary for $m = \{6, 12\},\$ $\varepsilon_{c} = \{-1.1, -0.9\}$. One can see that the modes are locally exponentially decreasing moving away from the interface (and oscillatory along the interface in $e^{im\theta}$), which is the mathematical characterization of surface plasmons [31, 8]. Going back to the scattering problem, it turns out that the dashed blue lines in Fig. 1 correspond to the real part of the interface resonances: $(\Re(\ell_m))_{m\geq 1}$ with $\ell_m^2 \in \mathcal{R}_{\mathsf{pla}}[-1.1, 1]$. In other words, the instabilities observed in the scattering problems are caused by plasmonic resonances close to the real axis. In the next section, we characterize this interface resonances family $(\ell_m)_{m\geq 1}$ by performing asymptotic expansion as $m\to\infty$. In particular we will confirm that $\overline{\ell}_m^2 \propto m^2$.

3.3. Interpretation with Schrdinger operator for the disk. From Section 3.2 we found that the nature of plasmonic resonances changes depending on ε_{c} (i.e. η). In this section we use asymptotic expansions to explain this change of behavior. To do so we provide an analogy with the Schrdinger operator. One can rewrite Problem Eq. (3.7) as

$$\begin{cases} -m^{-2}\frac{1}{r}\partial_r\left(r\,\partial_r w_m^{\pm}\right) + \frac{1}{r^2}w_m^{\pm} = \varepsilon\,\check{\lambda}w_m^{\pm} & \text{in }(0,R) \text{ and }(R,+\infty) \\ w_m^-(R) = w_m^+(R) \text{ and } -\eta^{-2}\partial_r w_m^-(R) = \partial_r w_m^+(R) & \text{across }\{R\} \\ w_0^{-\prime}(0) = 0 \text{ and } w_m^+ \in \mathscr{S}([R,+\infty)) \end{cases}$$

$$(3.11)$$

with $\check{\lambda} = m^{-2} \ell^2$ the new spectral parameter, w_m^{\pm} restrictions of w_m in each material, and $\mathscr{S}(\mathbb{R}_+)$ denoting the Schwartz space. We replace the outgoing wave condition by the Schwartz in order to characterize exponentially decreasing behaviors from both sides of the interface (a.k.a surface plasmons). To identify this behavior, first we rescale the problem Eq. (3.11) by $\xi = r/R - 1$ such that r = R corresponds to $\xi = 0$. We then

define
$$v_m^{\pm}(\xi) = w_m^{\pm}(R(1+\xi))$$
, and rewrite Problem Eq. (3.11) as

$$\begin{cases}
-m^{-2} \mathscr{L} v_m^{\pm} + V v_m^{\pm} = \varepsilon R^2 \breve{\lambda} v_m^{\pm} & \text{in } (-1,0) \text{ and } (1,+\infty) \\
v_m^{-}(0) = v_m^{+}(0) \text{ and } -\eta^{-2} \partial_{\xi} v_m^{-}(0) = \partial_{\xi} v_m^{+}(0) & \text{across } \{0\} \\
v_0^{-'}(-1) = 0 \text{ and } v_m^{+} \in \mathscr{S}(\mathbb{R}_+)
\end{cases}$$
(3.12)

where $\mathscr{L}(\xi, \partial_{\xi}) = \frac{1}{1+\xi} \partial_{\xi}((1+\xi) \partial_{\xi})$ is a positive elliptic operator (Laplacian like) and $V(\xi) = \frac{1}{(1+\xi)^2}$ is a potential. In that sense, the operator $v \mapsto (-m^{-2} \mathscr{L} + V)v$ in (3.12) can be interpreted as a Schrdinger operator. Consequently, depending on the sign of $\check{\lambda}$ one can find an eigenvalue or a resonance: $\check{\lambda} < 0$ will correspond to an eigenvalue, and $\check{\lambda} > 0$ will correspond to a resonance. Note that the choice to set $\check{\lambda} = m^{-2}\ell^2$ implies that we look for the leading order only. In other words we look for $\ell^2 \in \mathbb{R}$ at leading order. Figure 5 illustrates both situations. To construct localized modes at the



FIGURE 5. Graphs of the potential V and the spectral parameter $\check{\lambda}$: for $\check{\lambda} < 0$ (a), for $\check{\lambda} > 0$ (b). In case (a) $V - R^2 \check{\lambda} > 0$ leads to an eigenvalue, in case (b) $V(\xi) - R^2 \check{\lambda} < 0$ for $\xi > \xi_0$ leads to a resonance.

interface, we consider the principal part of $-m^2 \mathscr{L} + V$ with its coefficients frozen at $\xi = 0$, corresponding to $-m^{-2}\partial_{\xi}^2 + 1$. It is then natural to rescale by $\sigma = m\xi$, and the leading order behavior of Problem Eq. (3.12) becomes

$$\begin{cases} -\partial_{\sigma}^{2}\varphi^{-} + \varphi^{-} = -\eta^{2} R^{2} \check{\lambda} \varphi^{-} & \text{in } (-\infty, 0) \\ -\partial_{\sigma}^{2}\varphi^{+} + \varphi^{+} = R^{2} \check{\lambda} \varphi^{+} & \text{in } (0, +\infty) \\ \varphi^{-}(0) = \varphi^{+}(0) \quad \text{and} \quad \eta^{-2} \partial_{\sigma} \varphi^{-}(0) = \partial_{\sigma} \varphi^{+}(0) \quad \text{across } \{0\} \end{cases}.$$
(3.13)
$$\varphi^{\pm} \in \mathscr{S}(\mathbb{R}_{\pm})$$

with $\varphi^{\pm}(\sigma) = v_m^{\pm}(\xi)$. Note that the condition $v_m^{-}(-1) = \varphi^{-}(-m) = 0$ becomes $\varphi^{-} \in \mathscr{S}(\mathbb{R}_{-})$ to keep a localized behavior as $m \to +\infty$. Solutions of (3.13) are given by $(\check{\lambda}, \varphi^{\pm}) = (R^{-2}(1 - \eta^{-2}), e^{-\eta^{\pm 1} |\sigma|})$, where the modes are exponentially decreasing on both sides of the interface $\sigma = 0$. Back to the initial problem, the leading behavior corresponds to

$$\underline{\ell}_{m}^{2} = \frac{m^{2}}{R^{2}} \left(1 - \eta^{-2} \right), \quad \text{and} \quad \underline{w}_{m}^{\pm}(r) = \exp\left(-\eta^{\pm 1} m \left| \frac{r}{R} - 1 \right| \right).$$
(3.14)

We conclude:

- when $\varepsilon_{c} < -1$ ($\eta > 1$), surface plasmons are associated to $\ell^{2} > 0$ (at first order), which corresponds to a resonance;
- when $-1 < \hat{\varepsilon}_{c} < 0$ ($0 < \eta < 1$), surface plasmons are associated to $\ell^{2} < 0$ (at first order), which corresponds to an eigenvalue.

We have then characterized the asymptotic behavior of surface plasmons by building pairs $(\underline{\ell}_m^2, \underline{w}_m)_{m \ge 1}$. The obtained results match the observed behaviors in previous sections, and provide accurate predictions of the instabilities in the scattering problem for $\varepsilon_{c} < -1$.

The case of the circular cavity with constant ε_{c} is quite intuitive, and the leading order computations are explicit. In the next section we generalize the approach, to any order, for the general case: arbitrary shaped smooth boundary, and varying coefficients $\varepsilon_{c} \in \mathscr{C}^{\infty}(\Omega)$. To that aim, we will use semi-classical WKB (Wentzel-Kramers-Brillouin) expansions in a tubular neighborhood of the interface, and matched asymptotic expansions along the interface. The higher order terms allow to show a superalgebraic behavior of the peaks seen in Fig. 1, explaining the exponential increase asymptotically.

Remark 3.5. In this paper we focus on the construction of the pairs $(\underline{\ell}_m^2, \underline{w}_m)_{m\geq 1}$ to characterize the asymptotic behavior of surface plasmons. The proper justification of the observed instabilities in the scattering problem (corresponding to the real part of the resonances ℓ_m is not addressed in this paper. This involves technical details of spectral theory for problems with sign-changing coefficients, and we refer the reader to [17] for details. However we discuss the connection between the constructed pairs and the well-posedness constant $N_{\varepsilon,\rho}$ in Section 5.

4. Asymptotic for metamaterial cavities

To asymptotically characterize the interface resonances and their different behaviors in the general case, we construct a sequence of approximate solutions $(\underline{\lambda}_m, \underline{u}_m)_{m\geq 1}$ of Problem Eq. (2.5), called quasi-pair.

Definition 4.1. A quasi-pair for Problem Eq. (2.5) is formed by a sequence $(\underline{\lambda}_m)_{m\geq 1}$ of real numbers called quasi-resonances, and a sequence $(\underline{u}_m)_{m\geq 1}$ of complex valued functions called quasi-modes that satisfy the following conditions:

(1) For any $m \ge 1$, the functions \underline{u}_m are uniformly compactly supported and

 $\underline{u}_m \in D(P)$, with $\|\underline{u}_m\|_{L^2(\mathbb{R}^2)} = 1$;

(2) We have the following quasi-pair estimate as $m \to +\infty$,

$$\|P\underline{u}_m - \underline{\lambda}_m \,\underline{u}_m\|_{\mathrm{L}^2(\mathbb{R}^2)} = \mathcal{O}\left(m^{-\infty}\right),\tag{4.1}$$

with the notation $a_m = \mathcal{O}(m^{-\infty})$ to indicate that for all $N \in \mathbb{N}$, there exists $C_N > 0$ such that $|a_m| \leq C_N m^{-N}$, for all $m \geq 1$.

4.1. Statement. Recall $\Omega \subset \mathbb{R}^2$ be a cavity of smooth boundary Γ (see Section 2.1). Let L be the length of Γ , and we denote $\varepsilon_{c} = -\eta^{2}$ with $\eta \coloneqq \sqrt{-\varepsilon_{c}} \in \mathscr{C}^{\infty}(\overline{\Omega}, (0, +\infty))$ a positive smooth function.

Theorem 4.2. If $\eta|_{\Gamma} \neq 1$ ($\varepsilon_{\mathsf{c}}|_{\Gamma} \neq -1$), there exists a quasi-pair $(\underline{\lambda}_m, \underline{u}_m)_{m\geq 1}$ of Problem Eq. (2.5). Moreover, we have $\underline{\lambda}_m \propto m^2$, and

- $\underline{\lambda}_m > 0$, $\lim_{m \to \infty} \underline{\lambda}_m = +\infty$, if $\eta|_{\Gamma} > 1$; $\underline{\lambda}_m < 0$, $\lim_{m \to \infty} \underline{\lambda}_m = -\infty$, if $\eta|_{\Gamma} < 1$.

Remark 4.3. The proof of Theorem 4.2 relies on $\underline{\lambda}_m$ being constructed via a function $\Lambda \in$ $\mathscr{C}^{\infty}([0, \frac{L}{2\pi}])$ so that $\underline{\lambda}_m = \left(\frac{2\pi m}{L}\right)^2 \Lambda\left(\frac{L}{2\pi m}\right)$, with Λ (see Lemma 4.6 and Lemma 4.11) admitting the Taylor expansion

$$\Lambda(h) = \varsigma \left\langle \left| 1 - \frac{1}{\eta |_{\Gamma}^{2}} \right|^{-\frac{1}{2}} \right\rangle^{-2} + \mathcal{O}(h) \text{ as } h \to 0, \text{ with } \langle f \rangle \coloneqq \frac{1}{L} \int_{\Gamma} f \, \mathrm{d}\Gamma, \, \forall f \in \mathrm{L}^{1}(\Gamma).$$

Above, $\varsigma = \pm 1$ is the sign of $1 - \eta|_{\Gamma}^{-2}$ and $\langle \cdot \rangle$ is the mean along the interface. From the above construction, one can easily check the properties from $\underline{\lambda}_m$ stated in Theorem 4.2.

4.2. **Proof.** The proof of Theorem 4.2 follows the same structure as the proof of Theorem 4.A in [6], which asymptotically characterizes quasi-pairs for WGM in classic cavities. We make use of a different scaling to capture surface plasmons.

4.2.1. Formal expansions. First, we define a tubular neighborhood \mathcal{V}_{δ} of the interface Γ . Let $\gamma : \mathbb{T}_L \to \Gamma$ be a counterclockwise curvilinear parameterization of Γ with the notation $\mathbb{T}_L := \mathbb{R}/L\mathbb{Z}$. Let $n = (\gamma'_2, -\gamma'_1)^{\mathsf{T}}$ be the unit exterior normal to Ω and $\kappa : \mathbb{T}_L \to \mathbb{R}$ be the signed curvature. We define the tubular neighborhood $\mathcal{V}_{\delta} := \{\gamma(s) + \xi n(s) \mid (s,\xi) \in \mathbb{T}_L \times (-\delta, \delta)\}$ [33], see Fig. 6.



FIGURE 6. Tubular neighborhood and notations: s denotes an arclength parametrization of the curve γ , and ξ is the normal variable.

We now consider the problem:

$$\begin{cases} Pu = \lambda u & \text{in } \Omega \cap \mathcal{V}_{\delta} \text{ and } (\mathbb{R}^2 \setminus \overline{\Omega}) \cap \mathcal{V}_{\delta} \\ [u]_{\Gamma} = 0 \text{ and } [\varepsilon^{-1} \partial_n u]_{\Gamma} = 0 & \text{across } \Gamma \\ u = 0 & \text{on } \partial \mathcal{V}_{\delta} \end{cases}$$
(4.2)

where $P = -\operatorname{div}(\varepsilon^{-1}\nabla)$ with ε defined in Eq. (2.2). By Definition 4.1, the quasipairs are compactly supported therefore the outgoing condition does not play a role in their construction. We replace in particular the outgoing wave condition by an homogeneous Dirichlet boundary condition in order to construct localized quasi-pairs. The formal construction relies on: (i) an initialization where we setup the equation and the expansions in the tubular neighborhood, (ii) the leading order term computation, and (iii) a recurrence to compute higher order terms.

Initialization. The change of variables from the tubular coordinates $(s,\xi) \in \mathbb{T}_L \times (-\delta, \delta)$ to the Cartesian coordinates $x \in \mathcal{V}_{\delta}$ is a smooth diffeomorphism for $0 < \delta < (\max_{\mathbb{T}_L} |\kappa|)^{-1}$. In this tubular coordinate system the operator P becomes

$$P = -g^{-1} \operatorname{div}_{s,\xi} \left(\varepsilon^{-1} G \nabla_{s,\xi} \right)$$
where $g(s,\xi) = 1 + \xi \kappa(s) > 0$ and $G(s,\xi) = \begin{pmatrix} g(s,\xi)^{-1} & 0 \\ 0 & g(s,\xi) \end{pmatrix}$.
$$(4.3)$$

For the general case we use a WKB framework [5] in order to provide an asymptotic expansion of the spectral parameter as the number of oscillations around the interface,

denoted m in Section 3.3, goes to infinity. We introduce a small parameter h > 0 that will later be linked to m and the ansatz for the quasi-pair (λ, u) :

$$u(s,\xi) = w(s,\xi) \exp\left(\frac{i}{h}\theta(s)\right) \text{ and } \lambda = h^{-2}\breve{\lambda}$$
 (4.4)

where $w : \mathbb{T}_L \times (-\delta, \delta) \to \mathbb{C}, \ \theta : [0, L) \to \mathbb{C}$, and $\check{\lambda} \in \mathbb{C}$. We add the constrain $s \mapsto e^{\frac{i}{\hbar}\theta(s)} \in \mathscr{C}^{\infty}(\mathbb{T}_L)$ so that the function u in Eq. (4.4) is a smooth function in $\mathcal{V}_{\delta} \setminus \Gamma$. Following [5] we formally expand the unknowns w, θ , and $\check{\lambda}$ as

$$w(s,\xi) = \sum_{n\geq 0} w_n(s,\xi) \ h^n, \quad \theta(s) = \sum_{n\geq 0} \theta_n(s) \ h^n, \quad \text{and} \quad \breve{\lambda} = \sum_{n\geq 0} \breve{\lambda}_n \ h^n.$$
(4.5)

System Eq. (4.2) with the new unknowns Eq. (4.4) becomes

$$\begin{cases} \mathcal{L}_{h}[\varepsilon](w,\theta) = \breve{\lambda} w & \text{in } \mathbb{T}_{L} \times ((-\delta,\delta) \setminus \{0\}) \\ [w]_{\mathbb{T}_{L} \times \{0\}} = 0 \text{ and } [\varepsilon^{-1} \partial_{\xi} w]_{\mathbb{T}_{L} \times \{0\}} = 0 & \text{across } \mathbb{T}_{L} \times \{0\} \\ w = 0 & \text{on } \mathbb{T}_{L} \times \{-\delta,\delta\} \end{cases}$$
(4.6)

Above, $\mathcal{L}_h[\varepsilon](w,\theta) = h^2 e^{-\frac{i}{h}\theta} P(w e^{\frac{i}{h}\theta})$, and it can be decomposed as $\mathcal{L}_h[\varepsilon](w,\theta) = \mathcal{L}_h^3[\varepsilon](w,\theta,\theta) + \mathcal{L}_h^2[\varepsilon](w,\theta) + \mathcal{L}_h^4[\varepsilon](w)$

$$\mathcal{L}_h[\varepsilon](w,\theta) = \mathcal{L}_h^3[\varepsilon](w,\theta,\theta) + \mathcal{L}_h^2[\varepsilon](w,\theta) + \mathcal{L}_h^1[\varepsilon](w)$$
(4.7)

where $\mathcal{L}_{h}^{j}[\varepsilon]$ are *j*-linear for $j \in \{1, 2, 3\}$ and

$$\mathcal{L}_{h}^{3}[\varepsilon](w,\theta,\vartheta) = g^{-2} \varepsilon^{-1} w \,\partial_{s}\theta \,\partial_{s}\vartheta, \qquad (4.8a)$$

$$\mathcal{L}_{h}^{2}[\varepsilon](w,\theta) = -h \operatorname{i}\left(g^{-2} \varepsilon^{-1} \partial_{s} w \,\partial_{s} \theta + g^{-1} \,\partial_{s}\left(g^{-1} \varepsilon^{-1} \,w \,\partial_{s} \theta\right)\right), \tag{4.8b}$$

$$\mathcal{L}_{h}^{1}[\varepsilon](w) = -h^{2} g^{-1} \left(\partial_{\xi} \left(g \varepsilon^{-1} \partial_{\xi} w \right) + \partial_{s} \left(g^{-1} \varepsilon^{-1} \partial_{s} w \right) \right).$$
(4.8c)

In the above decomposition, only $\mathcal{L}_{h}^{1}[\varepsilon]$ involves derivatives with respect to ξ . Since g (resp. $\eta = \sqrt{-\varepsilon_{c}} > 0$) is a smooth function on $\mathbb{T}_{L} \times (-\delta, \delta)$ (resp. $\mathbb{T}_{L} \times (-\delta, 0]$), then G is smooth and we write the formal Taylor expansions about $\xi = 0$:

$$g(s,\xi) = 1 + \xi \kappa(s), \quad G(s,\xi) = \sum_{n \ge 0} \frac{\partial_{\xi}^{n} G(s,0)}{n!} \xi^{n}, \quad \eta(s,\xi) = \sum_{n \ge 0} \frac{\eta_{n}(s)}{n!} \xi^{n}, \quad (4.9)$$

where $\eta_n(s) = \partial_{\xi}^n \eta(s, 0)$. Since g and η do not vanish on $\mathbb{T}_L \times \{0\}$, the formal expansions of g^{-1} , g^{-2} , and η^{-2} about $\xi = 0$ can be computed with Eq. (4.9).

Like in Section 3.3, we introduce the scaled variable $\sigma = h^{-1}\xi$ for the normal variable $\xi \in (-\delta, \delta)$, and we define

$$\varphi^{\pm}(s,\sigma) = w(s,h\sigma) \quad \text{for } (s,\sigma) \in \mathbb{T}_L \times \mathbb{R}_{\pm}.$$
 (4.10)

Then with $g = g(s, h\sigma)$ we rewrite

$$\mathcal{L}_{h}^{1}[\varepsilon](\varphi^{\pm}) = -g^{-1} \partial_{\sigma} \left(\varepsilon^{-1} g \partial_{\sigma} \varphi^{\pm}\right) - h^{2} g^{-1} \partial_{s} \left(\varepsilon^{-1} g^{-1} \partial_{s} \varphi^{\pm}\right).$$
(4.11)

Problem Eq. (4.6) becomes the formal problem: Find $(\varphi_n^{\pm})_{n\in\mathbb{N}} \in \mathscr{C}^{\infty}(\mathbb{T}_L, \mathscr{S}(\mathbb{R}_{\pm}))^{\mathbb{N}}$, $(\exp(i h^{-1} \theta_n))_{n\in\mathbb{N}} \in \mathscr{C}^{\infty}(\mathbb{T}_L)^{\mathbb{N}}$, and $(\check{\lambda}_n)_{n\in\mathbb{N}} \in \mathbb{C}^{\mathbb{N}}$ such that

$$\begin{cases} \mathcal{L}_{h}[\varepsilon] \left(\sum_{n} \varphi_{n}^{\pm} h^{n}, \sum_{n} \theta_{n} h^{n}\right) = \left(\sum_{n} \breve{\lambda}_{n} h^{n}\right) \left(\sum_{n} \varphi_{n}^{\pm} h^{n}\right) & \text{in } \mathbb{T}_{L} \times \mathbb{R}_{\pm}^{*} \\ \sum_{n} \varphi_{n}^{-}(s, 0) h^{n} = \sum_{n} \varphi_{n}^{+}(s, 0) h^{n} & \text{on } \mathbb{T}_{L} \times \{0\} \\ -\eta_{0}(s)^{-2} \sum_{n} \partial_{\sigma} \varphi_{n}^{-}(s, 0) h^{n} = \sum_{n} \partial_{\sigma} \varphi_{n}^{+}(s, 0) h^{n} & \text{on } \mathbb{T}_{L} \times \{0\} \end{cases}$$
(4.12)

Note that for simplicity we extend the scaled domain $\mathbb{T}_L \times (-\frac{\delta}{h}, \frac{\delta}{h})$ to the domain $\mathbb{T}_L \times \mathbb{R}$ in order to be independent of h in Eq. (4.12), and we replace the homogeneous Dirichlet boundary condition on $\mathbb{T}_L \times \{-\frac{\delta}{h}, \frac{\delta}{h}\}$ by the conditions $\sigma \mapsto \varphi^{\pm}(s, \sigma) \in \mathscr{S}(\mathbb{R}_{\pm})$ for all $s \in \mathbb{T}_L$. One can always multiply the quasi-mode by a cutoff function $\xi \mapsto \chi(\xi)$ to be in the domain $\mathbb{T}_L \times (-\frac{\delta}{h}, \frac{\delta}{h})$, as done later in Eq. (4.18). With Eq. (4.7) and Eq. (4.9), we can formally expand the operators $\mathcal{L}_h^j[-\eta^2] = \sum_{n\geq 0} \mathbf{L}_n^{j,-} h^n$ and $\mathcal{L}_h^j[1] = \sum_{n\geq 0} \mathbf{L}_n^{j,+} h^n$ where $\mathbf{L}_{n}^{j,\pm}$ are independent of h, for $j \in \{1, 2, 3\}$. From Problem Eq. (4.12) we obtain the family of problems $(\mathcal{P}_{n})_{n \in \mathbb{N}}$ by identifying powers of h:

$$\begin{cases} \sum_{p \in \mathbb{N}_{n}^{4}} \mathbf{L}_{p_{1}}^{3,\pm} \left(\varphi_{p_{2}}^{\pm}, \theta_{p_{3}}, \theta_{p_{4}} \right) + \sum_{p \in \mathbb{N}_{n}^{3}} \mathbf{L}_{p_{1}}^{2,\pm} \left(\varphi_{p_{2}}^{\pm}, \theta_{p_{3}} \right) + \sum_{p \in \mathbb{N}_{n}^{2}} \mathbf{L}_{p_{1}}^{1,\pm} \left(\varphi_{p_{2}}^{\pm} \right) = \sum_{p \in \mathbb{N}_{n}^{2}} \breve{\lambda}_{p_{1}} \varphi_{p_{2}}^{\pm} \\ \varphi_{p}^{-}(s, 0) = \varphi_{p}^{+}(s, 0) \quad \text{and} \quad -\eta_{0}(s)^{-2} \partial_{\sigma} \varphi_{p}^{-}(s, 0) = \partial_{\sigma} \varphi_{p}^{+}(s, 0) \end{cases}$$
(4.13)

 $\varphi_n(s,0) = \varphi_n^+(s,0) \quad \text{and} \quad -\eta_0(s)^{-2} \,\partial_\sigma \varphi_n^-(s,0) = \partial_\sigma \varphi_n^+(s,0)$ with the notation $\mathbb{N}_n^d = \{ p \in \mathbb{N}^d \mid p_1 + \dots + p_d = n \}.$

<

Leading order terms. From Eq. (4.7) and Eq. (4.9) one obtains

Lemma 4.4. The first terms of the expansions of $\mathcal{L}_{h}^{3}[\varepsilon]$, $\mathcal{L}_{h}^{2}[\varepsilon]$, and $\mathcal{L}_{h}^{1}[\varepsilon]$, are given by

$$\mathbf{L}_{0}^{3,-}(\phi,\theta,\vartheta) = -\eta_{0}^{-2} \phi \partial_{s}\theta \partial_{s}\vartheta, \qquad \mathbf{L}_{0}^{2,-}(\phi,\theta) = 0, \qquad \mathbf{L}_{0}^{1,-}(\phi) = \eta_{0}^{-2} \partial_{\sigma}^{2}\phi, \qquad (4.14a)$$
$$\mathbf{L}_{0}^{3,+}(\phi,\theta,\vartheta) = \phi \partial_{s}\theta \partial_{s}\vartheta, \qquad \mathbf{L}_{0}^{2,+}(\phi,\theta) = 0, \qquad \mathbf{L}_{0}^{1,+}(\phi) = -\partial_{\sigma}^{2}\phi. \qquad (4.14b)$$

$$\mathbf{L}_{0}^{3,+}(\phi,\theta,\vartheta) = \phi \,\partial_{s}\theta \,\partial_{s}\vartheta, \qquad \mathbf{L}_{0}^{2,+}(\phi,\theta) = 0, \qquad \mathbf{L}_{0}^{1,+}(\phi) = -\partial_{\sigma}^{2}\phi. \tag{4.14b}$$

Using Lemma 4.4, we rewrite Problem (\mathcal{P}_0) as: Find $\varphi_0^{\pm} \in \mathscr{C}^{\infty}(\mathbb{T}_L, \mathscr{S}(\mathbb{R}_{\pm})), \exp(\mathrm{i} h^{-1} \theta_0) \in \mathbb{C}^{\infty}(\mathbb{T}_L, \mathbb{C}(\mathbb{R}_{\pm}))$ $\mathscr{C}^{\infty}(\mathbb{T}_L)$, and $\check{\lambda}_0 \in \mathbb{C}$ such that $(\varphi_0^-, \varphi_0^+) \neq (0, 0)$ and

$$\begin{cases} \partial_{\sigma}^{2}\varphi_{0}^{-} - \left(\theta_{0}^{\prime 2} + \eta_{0}^{2}\breve{\lambda}_{0}\right)\varphi_{0}^{-} = 0 & \text{ in } \mathbb{T}_{L} \times \mathbb{R}_{-} \\ \partial_{\sigma}^{2}\varphi_{0}^{+} - \left(\theta_{0}^{\prime 2} - \breve{\lambda}_{0}\right)\varphi_{0}^{+} = 0 & \text{ in } \mathbb{T}_{L} \times \mathbb{R}_{+} \\ \varphi_{0}^{-}(s,0) = \varphi_{0}^{+}(s,0) & \text{ on } \mathbb{T}_{L} \times \{0\} \\ -\eta_{0}(s)^{-2}\partial_{\sigma}\varphi_{0}^{-}(s,0) = \partial_{\sigma}\varphi_{0}^{+}(s,0) & \text{ on } \mathbb{T}_{L} \times \{0\} \end{cases}$$

$$(4.15)$$

Notation 4.5. We recall that $\eta_0 = \eta|_{\Gamma}$ and, we define the scalar $\varsigma = \pm 1$ to be the sign of $1 - \eta_0^{-2}$, the functions $\tau_0 = |1 - \eta_0^{-2}|^{-\frac{1}{2}}$ and $\hat{\tau}_0 = \frac{\tau_0}{\langle \tau_0 \rangle}$.

Lemma 4.6. One can choose $h = \frac{L}{2\pi m}$ for $m \in \mathbb{N}^*$ so that $(\varphi_0^{\pm}, \theta_0, \check{\lambda}_0)$ given by $\check{\lambda}_0 = \frac{\varsigma}{\langle \tau_0 \rangle^2}, \quad \theta_0(s) = \int_0^s \widehat{\tau}_0(t) \, \mathrm{d}t, \quad and \quad \varphi_0^{\pm}(s,\sigma) = \alpha(s) \exp\left(-|\sigma| \, \widehat{\tau}_0(s) \, \eta_0(s)^{\mp 1}\right),$

with $\alpha \in \mathscr{C}^{\infty}(\mathbb{T}_L, \mathbb{C}^*)$, is solution of Problem (\mathcal{P}_0) defined in Eq. (4.15).

The proof is detailed in Appendix B.1.

Remark 4.7. If we unravel the scaling and return to tubular coordinates, for $m \geq 1$ and $(s,\xi) \in \mathbb{T}_L \times \mathbb{R}$, we formally have a pair $(\underline{\lambda}_m, \underline{u}_m)$

$$\underline{\lambda}_{m} = \left(\frac{2\pi m}{L}\right)^{2} \left[\breve{\lambda}_{0} + \mathcal{O}\left(m^{-1}\right)\right],$$

$$\underline{u}_{m}(s,\xi) = e^{i\frac{2\pi m}{L} \left[\theta_{0}(s) + \mathcal{O}\left(m^{-1}\right)\right]} \left[\begin{cases} \varphi_{0}^{-}\left(s,\frac{2\pi m}{L}\xi\right) & \text{if } \xi \leq 0\\ \varphi_{0}^{+}\left(s,\frac{2\pi m}{L}\xi\right) & \text{if } \xi > 0 \end{cases} + \mathcal{O}\left(m^{-1}\right) \end{bmatrix},$$

which characterize surface plasmons at leading order.

Remark 4.8. The construction relies on several choices that are not unique.

- One can choose the main phase to satisfy $\theta'_0 = \hat{\tau}_0$ or $\theta'_0 = -\hat{\tau}_0$. Then one can construct two modes corresponding to \underline{u}_m and $\overline{\underline{u}_m}$ (see Remark 4.16).
- The function θ_0 is defined up to a constant c. Then \underline{u}_m in Remark 4.7 is defined up to $e^{i\frac{2\pi m}{L}c}$. For simplicity we consider c = 0 as we normalize in the end.
- The functions φ_0^{\pm} are defined up to a function $\alpha : \mathbb{T}_L \to \mathbb{C}^*$, which contributes to the phase of \underline{u}_m and therefore affects the number of oscillations along the interface. One can always shift indices so that $(\underline{\lambda}_m, \underline{u}_m)_{m \ge 1-q_\alpha}$, for some $q_\alpha \in \mathbb{Z}$, corresponds to a wave with m oscillations along the interface.

Recurrence. From Eq. (4.13), Lemma 4.4, and Lemma 4.6, for $n \ge 1$, we can rewrite Problem (\mathcal{P}_n) as: Find $\varphi_n^{\pm} \in \mathscr{C}^{\infty}(\mathbb{T}_L, \mathscr{S}(\mathbb{R}_{\pm}))$, $\exp(i h^{n-1} \theta_n) \in \mathscr{C}^{\infty}(\mathbb{T}_L)$, and $\check{\lambda}_n \in \mathbb{C}$ such that

$$\begin{cases} \partial_{\sigma}^{2}\varphi_{n}^{-} - \hat{\tau}_{0}^{2}\eta_{0}^{2}\varphi_{n}^{-} = \left(2\hat{\tau}_{0}\theta_{n}' + \eta_{0}^{2}\breve{\lambda}_{n}\right)\varphi_{0}^{-} + \eta_{0}^{2}S_{n-1}^{-} & \text{in } \mathbb{T}_{L} \times \mathbb{R}_{-} \\ \partial_{\sigma}^{2}\varphi_{n}^{+} - \hat{\tau}_{0}^{2}\eta_{0}^{-2}\varphi_{n}^{+} = \left(2\hat{\tau}_{0}\theta_{n}' - \breve{\lambda}_{n}\right)\varphi_{0}^{+} - S_{n-1}^{+} & \text{in } \mathbb{T}_{L} \times \mathbb{R}_{+} \\ \varphi_{n}^{-}(s,0) = \varphi_{n}^{+}(s,0) & \text{on } \mathbb{T}_{L} \times \{0\} \\ -\eta_{0}(s)^{-2}\partial_{\sigma}\varphi_{n}^{-}(s,0) = \partial_{\sigma}\varphi_{n}^{+}(s,0) & \text{on } \mathbb{T}_{L} \times \{0\} \end{cases}$$

$$(4.16)$$

where

$$S_{n-1}^{\pm} = \sum_{p=1}^{n-1} \breve{\lambda}_{n-p} \, \varphi_p^{\pm} - \sum_{p=0}^{n-1} \mathbf{L}_{n-p}^{1,\pm} \left(\varphi_p^{\pm} \right) - \sum_{p \in \mathbb{N}_{n-1}^3} \mathbf{L}_{p_1}^{2,\pm} \left(\varphi_{p_2}^{\pm}, \theta_{p_3} \right) \\ - \mathbf{L}_n^{2,\pm} \left(\varphi_0^{\pm}, \theta_0 \right) - \sum_{p \in \mathbb{N}_{n-1}^4} \mathbf{L}_{p_1}^{3,\pm} \left(\varphi_{p_2}^{\pm}, \theta_{p_3}, \theta_{p_4} \right) - \mathbf{L}_n^{3,\pm} \left(\varphi_0^{\pm}, \theta_0, \theta_0 \right). \quad (4.17)$$

Lemma 4.9. Define $(\varphi_0^{\pm}, \theta_0, \check{\lambda}_0)$ according to Lemma 4.6. For $n \geq 1$, there exists $(\varphi_n^{\pm}, \theta_n, \check{\lambda}_n) \in \mathscr{C}^{\infty}(\mathbb{T}_L, \mathscr{S}(\mathbb{R}_{\pm})) \times \mathscr{C}^{\infty}(\mathbb{T}_L) \times \mathbb{C}$ solution of Problem (\mathcal{P}_n) defined in Eq. (4.16). In particular, φ_n^{\pm} is given by

$$\varphi_n^{\pm}(s,\sigma) = P_n^{\pm}(s,\sigma) \exp\left(-|\sigma|\,\widehat{\tau}_0(s)\,\eta_0(s)^{\mp 1}\right),\,$$

with polynomials $P_n^{\pm} \in \mathscr{C}^{\infty}(\mathbb{T}_L, \mathbb{P})^1$.

The proof is detailed in Appendix B.2.

Remark 4.10. In addition to Remark 4.8, $(\theta_n)_{n\geq 0}$ and $(\varphi_n)_{n\geq 0}$ are not uniquely defined at each step of the construction. However, the sequence $(\check{\lambda}_n)_{n\geq 0}$ will unique in the sense of Corollary 4.15.

4.2.2. Quasi-pairs. Based on the formal series $\sum_{n \in \mathbb{N}} \varphi_n^{\pm} h^n$, $\sum_{n \in \mathbb{N}} \theta_n h^n$, and $\sum_{n \in \mathbb{N}} \check{\lambda}_n h^n$ with $h = \frac{L}{2\pi m}$, we now construct quasi-pairs in the sense of Definition 4.1. First we use Borel's Lemma [29, Thm. 1.2.6] for $\check{\lambda}$ and θ , and a direct generalization on the Fréchet space $\mathscr{C}^{\infty}(\mathbb{T}_L, \mathscr{S}(\mathbb{R}_{\pm}))$ [6, Lem. A.5] for φ^{\pm} to establish the

¹We denote \mathbb{P} the space of polynomial of a single variable with complex coefficients.

Lemma 4.11. There exists $\Phi^{\pm} \in \mathscr{C}^{\infty}([0, \frac{L}{2\pi}] \times \mathbb{T}_L, \mathscr{S}(\mathbb{R}_{\pm})), \Theta \in \mathscr{C}^{\infty}([0, \frac{L}{2\pi}] \times \mathbb{T}_L),$ and $\Lambda \in \mathscr{C}^{\infty}([0, \frac{L}{2\pi}])$ such that, for $N \geq 1$, $h \in [0, \frac{L}{2\pi}]$, $s \in \mathbb{T}_L$, and $\sigma \in \mathbb{R}_{\pm}$, we have

$$\Phi^{\pm}(h;s,\sigma) = \sum_{n=0}^{N-1} \varphi_n^{\pm}(s,\sigma) h^n + h^N R_N^{\pm}(h;s,\sigma),$$

$$\Theta(h;s) = \sum_{n=0}^{N-1} \theta_n(s) h^n + h^N R_N^{\Theta}(h;s), \quad and \quad \Lambda(h) = \sum_{n=0}^{N-1} \breve{\lambda}_n h^n + h^N R_N^{\Lambda}(h)$$

$$\pi R^{\pm} \in \mathscr{C}^{\infty}([0, L] \times \mathbb{T}_{\tau}, \mathscr{C}(\mathbb{P}_{\tau})), \quad R^{\Theta} \in \mathscr{C}^{\infty}([0, L] \times \mathbb{T}_{\tau}), \quad R^{\Lambda} \in \mathscr{C}^{\infty}([0, L]).$$

where $R_N^{\pm} \in \mathscr{C}^{\infty}([0, \frac{L}{2\pi}] \times \mathbb{T}_L, \mathscr{S}(\mathbb{R}_{\pm})), R_N^{\Theta} \in \mathscr{C}^{\infty}([0, \frac{L}{2\pi}] \times \mathbb{T}_L), R_N^{\Lambda} \in \mathscr{C}^{\infty}([0, \frac{L}{2\pi}]).$

From those functions, we now define the quasi-resonances $\underline{\lambda}_m$ and the quasi-modes \underline{u}_m in the tubular neighborhood as

$$\underline{\lambda}_{m} = \left(\frac{2\pi m}{L}\right)^{2} \Lambda\left(\frac{L}{2\pi m}\right)$$

$$\underline{u}_{m}(s,\xi) = \chi(\xi) \exp\left(\mathrm{i}\frac{2\pi m}{L}\Theta\left(\frac{L}{2\pi m};s\right)\right) \begin{cases} \Phi^{-}\left(\frac{L}{2\pi m};s,\frac{2\pi m}{L}\xi\right) & \text{if } \xi \leq 0 \\ \Phi^{+}\left(\frac{L}{2\pi m};s,\frac{2\pi m}{L}\xi\right) & \text{if } \xi > 0 \end{cases}$$
(4.18)

where χ is a cutoff function, $\chi \in \mathscr{C}^{\infty}_{\text{comp}}((-\delta, \delta))$ and $\chi \equiv 1$ on $\left[-\frac{\delta}{2}, \frac{\delta}{2}\right]$. In what follows, we establish that Eq. (4.18) is a quasi-pair. First we have

Lemma 4.12. The pair $(\underline{\lambda}_m, \underline{u}_m)_{m\geq 1}$ defined in Eq. (4.18) satisfies the following:

- (i) \underline{u}_m is uniformly compactly supported and smooth in Ω and $\mathbb{R}^2 \setminus \overline{\Omega}$.
- (ii) \underline{u}_m satisfies $[\underline{u}_m]_{\Gamma} = \mathcal{O}(m^{-\infty})$ and $[\varepsilon^{-1}\partial_n\underline{u}_m]_{\Gamma} = \mathcal{O}(m^{-\infty}).$
- (iii) \underline{u}_m admits the norm expansion $\|\underline{u}_m\|_{L^2(\mathbb{R}^2)} = a m^{-\frac{1}{2}} + \mathcal{O}(m^{-\frac{3}{2}}), a > 0.$
- (iv) Let $\underline{R}_m \coloneqq P \underline{u}_m \underline{\lambda}_m \underline{u}_m$, then $\|\underline{R}_m\|_{L^2(\Omega)} + \|\underline{R}_m\|_{L^2(\mathbb{R}^2 \setminus \overline{\Omega})} = \mathcal{O}(m^{-\infty})$.

(v) If two quasi-pairs $(\underline{\lambda}_m, \underline{u}_m)_{m \ge 1}$, $(\underline{\mu}_m, \underline{v}_m)_{m \ge 1}$ satisfy (i)-(iv), and the quasimodes have the same leading phase $\theta_0(s) = \int_0^s \widehat{\tau}_0(t) \, dt$ then: $\int_{\mathbb{R}^2} \underline{u}_m \, \overline{\underline{v}_m} \, dx = z_0 \, m^{-1} + \mathcal{O}(m^{-2}), \ z_0 \in \mathbb{C}^*, \ and \ \int_{\mathbb{R}^2} \underline{u}_m \, \underline{v}_m \, dx = \mathcal{O}(m^{-\infty}).$

Remark 4.13. Items (iii) and (v) of Lemma 4.12 give us

$$\int_{\mathbb{R}^2} \frac{\underline{u}_m}{\|\underline{u}_m\|_{\mathrm{L}^2(\mathbb{R}^2)}} \frac{\overline{\underline{v}_m}}{\|\overline{\underline{v}_m}\|_{\mathrm{L}^2(\mathbb{R}^2)}} \,\mathrm{d}x = z_0' + \mathcal{O}(m^{-1}), \quad \text{with } z_0' \in \mathbb{C}^*.$$

Proof. Recall that we set $h = \frac{L}{2\pi m}$, and to simplify notations we denote $\chi_h : \sigma \mapsto \chi(\sigma h)$, $\Phi_h^{\pm} : (s, \sigma) \mapsto \Phi^{\pm}(h; s, \sigma), \Theta_h : s \mapsto \Theta(h; s)$, and $\Lambda_h = \Lambda(h)$.

(i) By definition of $(\underline{u}_m)_{m\geq 1}$, (i) is satisfied.

(*ii*) Using Lemma 4.11, one can show that $[\underline{u}_m]_{\Gamma} = \mathcal{O}(m^{-N})$ and $[\varepsilon^{-1}\partial_n\underline{u}_m]_{\Gamma} = \mathcal{O}(m^{-N})$ for all $N \ge 0$, which is the definition of $\mathcal{O}(m^{-\infty})$.

(*iii*) We introduce the weighted L² semi-norm on $\mathbb{T}_L \times \mathbb{R}_{\pm}$

$$\|f\|_{\mathrm{L}^{2}_{\pm}[h]}^{2} = \int_{\mathbb{T}_{L}} \int_{\mathbb{R}_{\pm}\cap(-\frac{\delta}{h},\frac{\delta}{h})} |f(s,\sigma)|^{2} h(1+\kappa(s)\sigma h) \,\mathrm{d}\sigma \,\mathrm{d}s.$$
(4.19)

Form Eq. (4.18), we obtain

$$\left\|\underline{u}_{m}\right\|_{\mathrm{L}^{2}(\mathbb{R}^{2})}^{2} = \left\|\chi_{h}\Phi_{h}^{-}\mathsf{e}^{\frac{\mathrm{i}}{h}\Theta_{h}}\right\|_{\mathrm{L}^{2}_{-}[h]}^{2} + \left\|\chi_{h}\Phi_{h}^{+}\mathsf{e}^{\frac{\mathrm{i}}{h}\Theta_{h}}\right\|_{\mathrm{L}^{2}_{+}[h]}^{2}$$

From Lemma 4.6 and Lemma 4.11 for N = 1, we have

$$\Theta_h(s) = \int_0^s \tau_0(t) \, \mathrm{d}t + \theta_1(s)h + h^2 R_2^{\Theta}(h;s)$$

$$\Phi_h^{\pm}(s,\sigma) = \alpha(s) \, \exp\left(-|\sigma|\,\hat{\tau}_0(s)\,\eta_0(s)^{\pm 1}\right) + h \, R_1^{\pm}(h;s,\sigma)$$

where $R_2^{\Theta} \in \mathscr{C}^{\infty}([0, \frac{L}{2\pi}] \times \mathbb{T}_L)$ and $R_1^{\pm} \in \mathscr{C}^{\infty}([0, \frac{L}{2\pi}] \times \mathbb{T}_L, \mathscr{S}(\mathbb{R}_{\pm}))$. We deduce that $\left\| \left\| \chi_h \Phi_h^{\pm} \mathbf{e}^{\frac{i}{h} \Theta_h} \right\|_{\mathbf{L}^2_{\pm}[h]}^2 - \left\| \chi_h \alpha \, \mathbf{e}^{-|\sigma| \, \widehat{\tau}_0 \eta_0^{\pm 1}} \mathbf{e}^{\mathbf{i} \, \theta_1} \right\|_{\mathbf{L}^2_{\pm}[h]}^2 \right\| \leq C_1^{\pm} h^2$

$$\left\|\chi_{h}\Phi_{h}^{\pm}\mathsf{e}^{\frac{\mathrm{i}}{h}\Theta_{h}}\right\|_{\mathrm{L}^{2}_{\pm}[h]}^{2} - \left\|\chi_{h}\,\alpha\,\mathsf{e}^{-|\sigma|\,\widehat{\tau}_{0}\eta_{0}^{\pm1}}\,\mathsf{e}^{\mathrm{i}\,\theta_{1}}\right\|_{\mathrm{L}^{2}_{\pm}[h]}^{2} \le C_{1}^{\pm}h^{2}$$

for C_1^{\pm} some positive constant. We write $\left\|\chi_h \alpha \, \mathrm{e}^{-|\sigma|\,\widehat{\tau}_0\eta_0^{\pm 1}} \mathrm{e}^{\mathrm{i}\,\theta_1}\right\|_{\mathrm{L}^2_+[h]}^2 = I_1^{\pm} + I_2^{\pm} + I_3^{\pm},$

$$\begin{split} I_{1}^{\pm} &= h \int_{\mathbb{T}_{L}} \int_{\mathbb{R}_{\pm}} |\alpha(s)|^{2} \mathrm{e}^{\mp 2\sigma \, \widehat{\tau}_{0}(s)\eta_{0}(s)^{\mp 1}} \mathrm{e}^{-2\Im\theta_{1}(s)} \, \mathrm{d}\sigma \, \mathrm{d}s = h \int_{\mathbb{T}_{L}} \frac{|\alpha(s)|^{2} \mathrm{e}^{-2\Im\theta_{1}(s)}}{2\widehat{\tau}_{0}(s)\eta_{0}(s)^{\mp 1}} \, \mathrm{d}s, \\ I_{2}^{\pm} &= h \int_{\mathbb{T}_{L}} \int_{\mathbb{R}_{\pm}} (|\chi(\sigma h)|^{2} - 1) |\alpha(s)|^{2} \mathrm{e}^{\mp 2\sigma \, \widehat{\tau}_{0}(s)\eta_{0}(s)^{\mp 1}} \mathrm{e}^{-2\Im\theta_{1}(s)} \, \mathrm{d}\sigma \, \mathrm{d}s, \\ I_{3}^{\pm} &= h^{2} \int_{\mathbb{T}_{L}} \int_{\mathbb{R}_{\pm}} |\chi(\sigma h)\alpha(s)|^{2} \mathrm{e}^{\mp 2\sigma \, \widehat{\tau}_{0}(s)\eta_{0}(s)^{\mp 1}} \mathrm{e}^{-2\Im\theta_{1}(s)} \, \kappa(s)\sigma \, \mathrm{d}\sigma \, \mathrm{d}s. \end{split}$$

One can show that $I_2^{\pm} = \mathcal{O}(h^{\infty})$ using Lemma B.1. Since χ is bounded and the function $(h; s, \sigma) \mapsto |\alpha|^2 e^{\mp 2\sigma \hat{\tau}_0 \eta_0^{\pm 1}} e^{-2\Im \theta_1} \kappa \sigma$ is in $\mathscr{C}^{\infty}([0, \frac{L}{2\pi}] \times \mathbb{T}_L, \mathscr{S}(\mathbb{R}_{\pm}))$ there exists a constant C_3^{\pm} such that $|I_3^{\pm}| \leq C_3^{\pm} h^2$. Combining the results we get

$$\|\underline{u}_m\|_{\mathrm{L}^2(\mathbb{R}^2)}^2 = a^2 m^{-1} + \mathcal{O}(m^{-2})$$

with

$$a^2 = \frac{2\pi}{hL}(I_1^+ + I_1^-) = \frac{2\pi}{L} \int_{\mathbb{T}_L} |\alpha(s)|^2 \, \mathrm{e}^{-2\Im(\theta_1(s))} \, \frac{\eta_0(s)^{-1} + \eta_0(s)}{2\widehat{\tau}_0(s)} \, \mathrm{d}s > 0.$$

(iv) Revisiting the change of variables in tubular coordinates and the scaling, we get

$$\|\underline{R}_{m}\|_{\mathrm{L}^{2}(\Omega)} = h^{-2} \left\| \mathsf{e}^{\mathsf{i} h^{-1} \Theta_{h}} \left(\mathcal{L}_{h}[\varepsilon](\cdot, \Theta_{h}) - \Lambda_{h} \right) \left(\chi_{h} \Phi_{h}^{-} \right) \right\|_{\mathrm{L}^{2}_{-}[h]}, \tag{4.20a}$$

$$\|\underline{R}_{m}\|_{\mathrm{L}^{2}(\mathbb{R}^{2}\setminus\overline{\Omega})} = h^{-2} \left\| \mathsf{e}^{\mathsf{i}\,h^{-1}\,\Theta_{h}} \left(\mathcal{L}_{h}[\varepsilon](\cdot,\Theta_{h}) - \Lambda_{h} \right) \left(\chi_{h}\,\Phi_{h}^{+} \right) \right\|_{\mathrm{L}^{2}_{+}[h]} \tag{4.20b}$$

with $\mathcal{L}_h[\varepsilon]$ defined in Eq. (4.6). Lemma 4.11 with N = 1 and Lemma 4.6 give the estimation $\Im \Theta_h = \mathcal{O}(h)$ so there exists $c_{\Theta} > 0$ such that $|e^{ih^{-1}\Theta_h}| \leq c_{\Theta}$. Introducing the commutator $[\mathcal{L}_h[\varepsilon](\cdot,\Theta_h),\chi_h]$ of the differential operator $\Phi\mapsto\mathcal{L}_h[\varepsilon](\Phi,\Theta_h)$ with the scaled cutoff function χ_h , we deduce from Eq. (4.20)

 $\|\underline{R}_m\|_{\mathrm{L}^2(\Omega)} \le c_{\Theta} h^{-2} \left(\mathcal{N}_- + \mathcal{N}_-'\right) \quad \text{and} \quad \|\underline{R}_m\|_{\mathrm{L}^2(\mathbb{R}^2 \setminus \overline{\Omega})} \le c_{\Theta} h^{-2} \left(\mathcal{N}_+ + \mathcal{N}_+'\right) \quad (4.21)$ where $\mathcal{N}_{\pm} = \left\| \chi_h \left(\mathcal{L}_h[\varepsilon](\cdot, \Theta_h) - \Lambda_h \right) \Phi_h^{\pm} \right\|_{\mathrm{L}^2_{\pm}[h]}, \mathcal{N}'_{\pm} = \left\| \left[\mathcal{L}_h[\varepsilon](\cdot, \Theta_h), \chi_h \right] \Phi_h^{\pm} \right\|_{\mathrm{L}^2_{\pm}[h]}.$ Let's start with \mathcal{N}_{\pm} . We write for $N \geq 1$,

$$\mathcal{L}_{h}[\varepsilon](\Phi_{h}^{\pm},\Theta_{h}) = \sum_{n=0}^{N-1} h^{n} \left(\mathbf{L}_{n}^{\pm,3}(\Phi_{h}^{\pm},\Theta_{h},\Theta_{h}) + \mathbf{L}_{n}^{\pm,2}(\Phi_{h}^{\pm},\Theta_{h}) + \mathbf{L}_{n}^{\pm,1}(\Phi_{h}^{\pm}) \right) \\ + h^{N} \left(\mathbf{R}_{N}^{\pm,3}(h;\Phi_{h}^{\pm},\Theta_{h},\Theta_{h}) + \mathbf{R}_{N}^{\pm,2}(h;\Phi_{h}^{\pm},\Theta_{h}) + \mathbf{R}_{N}^{\pm,1}(h;\Phi_{h}^{\pm}) \right)$$

where $\mathbf{R}_N^{\pm,j}(h)$ are *j*-linear second order differential operators such that all the coefficients in $\chi_h \mathbf{R}_N^{\pm,j}(h)$ are smooth bounded functions for $j \in \{1, 2, 3\}$. We use Lemma 4.11 with different N for each occurrence of Φ_h^{\pm} and Θ_h , and we obtain

$$\mathcal{L}_{h}[\varepsilon](\Phi_{h}^{\pm},\Theta_{h}) - \Lambda_{h}\Phi_{h}^{\pm}$$

$$= h^{N} \left[\sum_{n=0}^{N-1} \sum_{p \in \mathbb{N}_{N-n}^{3}} \mathbf{L}_{n}^{\pm,3}(R_{p_{1}}^{\pm}(h), R_{p_{2}}^{\Theta}(h), R_{p_{3}}^{\Theta}(h)) + \mathbf{R}_{N}^{\pm,3}(h; R_{0}^{\pm}(h), R_{0}^{\Theta}(h), R_{0}^{\Theta}(h)) \right]$$

$$+ \sum_{n=0}^{N-1} \sum_{p \in \mathbb{N}_{N-n}^{2}} \mathbf{L}_{n}^{\pm,2}(R_{p_{1}}^{\pm}(h), R_{p_{2}}^{\Theta}(h)) + \mathbf{R}_{N}^{\pm,2}(h; R_{0}^{\pm}(h), R_{0}^{\Theta}(h))$$

$$+ \sum_{n=0}^{N-1} \left(\mathbf{L}_{n}^{\pm,1} - \breve{\lambda}_{n} \right) R_{N-n}^{\pm}(h) + \mathbf{R}_{N}^{\pm,1}(h; R_{0}^{\pm}(h)) - R_{N}^{\Lambda}(h) R_{0}^{\pm}(h) \right]$$

where we used the relations in Eq. (4.13), giving us that for all $Q \in \mathbb{N}$

$$\sum_{p \in \mathbb{N}_Q^4} \mathbf{L}_{p_1}^{3,\pm} \left(\varphi_{p_2}^{\pm}, \theta_{p_3}, \theta_{p_4} \right) + \sum_{p \in \mathbb{N}_Q^3} \mathbf{L}_{p_1}^{2,\pm} \left(\varphi_{p_2}^{\pm}, \theta_{p_3} \right) + \sum_{p \in \mathbb{N}_Q^2} \left(\mathbf{L}_{p_1}^{1,\pm} - \breve{\lambda}_{p_1} \right) \left(\varphi_{p_2}^{\pm} \right) = 0.$$

The coefficients in the operator $\chi_h \mathcal{L}_h[\varepsilon](\cdot, \Theta_h)$ are smooth bounded functions in $\mathbb{T}_L \times \mathbb{R}_\pm$ (see Eq. (4.8a), Eq. (4.8b), Eq. (4.11)). From Eq. (4.22), we get $\mathcal{N}_\pm \leq h^N \|F^\pm(h)\|_{L_\pm[h]}$ where $F^\pm \in \mathscr{C}^\infty([0, \frac{L}{2\pi}] \times \mathbb{T}_L, \mathscr{S}(\mathbb{R}_\pm))$ so we have $\mathcal{N}_\pm \leq C_N h^N$ for C_N a constant independent of h as $h \to 0$. Now, we consider the two commutator norms \mathcal{N}'_\pm . We observe that the coefficients of the operators $[\mathcal{L}_h[\varepsilon](\cdot, \Theta_h), \chi_h]$ are zero in $\mathbb{T}_L \times (-\frac{\delta}{2h}, 0)$ and $\mathbb{T}_L \times (0, \frac{\delta}{2h})$. From this observation, we deduce that

$$\mathcal{N}_{\pm}'^{2} = \int_{\mathbb{T}_{L}} \int_{I_{\pm}(h)} |G^{\pm}(h;s,\sigma)|^{2} \,\mathrm{d}\sigma \,\mathrm{d}s$$

where $G^{\pm} \in \mathscr{C}^{\infty}([0, \frac{L}{2\pi}] \times \mathbb{T}_L, \mathscr{S}(\mathbb{R}_{\pm}))$ and $I_{\pm}(h)$ are as in Lemma B.1 for $\rho = \frac{\delta}{2}$. We deduce that $\mathcal{N}'_{\pm} = \mathcal{O}(h^{\infty})$, and we get $\|\underline{R}_m\|_{L^2(\Omega)} + \|\underline{R}_m\|_{L^2(\mathbb{R}^2\setminus\overline{\Omega})} = \mathcal{O}(h^{N-2})$ for all N > 1.

(v) Let $(\theta_n)_{n\geq 0}$ (resp. $(\vartheta_n)_{n\geq 0}$) be a sequence of phases constructed for \underline{u}_m (resp. \underline{v}_m) and α (resp. β) the function in Lemma 4.6. A similar computation as in *(iii)* gives that $\int_{\mathbb{R}^2} \underline{u}_m \, \overline{v}_m \, \mathrm{d}x = z_0 \, h + \mathcal{O}(h^2)$ where

$$z_{0} = \sum_{\pm} \int_{\mathbb{T}_{L}} \alpha(s) \overline{\beta(s)} e^{i\theta_{1}(s) - i\overline{\vartheta_{1}(s)}} \int_{\mathbb{R}_{\pm}} e^{\mp 2\sigma \,\widehat{\tau}_{0}(s)\eta_{0}(s)^{\mp 1}} \,\mathrm{d}\sigma \,\mathrm{d}s$$
$$= \int_{\mathbb{T}_{L}} \alpha(s) \overline{\beta(s)} e^{i\theta_{1}(s) - i\overline{\vartheta_{1}(s)}} \,\frac{\eta_{0}(s)^{-1} + \eta_{0}(s)}{2\widehat{\tau}_{0}(s)} \,\mathrm{d}s.$$

Using Lemma B.2, we get $i\theta_1(s) - i\overline{\vartheta_1(s)} = -2f(s) - \int_0^s \frac{\alpha'(t)}{\alpha(t)} + \frac{\overline{\beta'(t)}}{\overline{\beta(t)}} dt$ where f is a real function independent of α and β . A derivative computation shows that the functions $\alpha e^{-\int \frac{\alpha'}{\alpha}} \equiv \alpha_0 \in \mathbb{C}^*, \ \beta e^{-\int \frac{\beta'}{\beta}} \equiv \beta_0 \in \mathbb{C}^*$ are constant so $z_0 = \alpha_0 \overline{\beta_0} \int_{\mathbb{T}_L} \frac{\eta_0(s)^{-1} + \eta_0(s)}{2\widehat{\tau_0(s)}} e^{-2f(s)} ds \neq 0$. Denoting R (resp. S) the remainder in the construction of \underline{u}_m (resp. \underline{v}_m), we have

$$\int_{\mathbb{R}^2} \underline{u}_m \, \underline{v}_m \, \mathrm{d}x = \int_{\mathbb{T}_L} F(h; s) \, \mathrm{e}^{\mathrm{i} \frac{4\pi m}{L} \theta_0(s)} \, \mathrm{d}s$$

where

$$F(h;s) = e^{iR_1^{\Theta}(h;s) + iS_1^{\Theta}(h;s)}$$
$$\sum_{\pm} \int_{\mathbb{R}_{\pm}} \chi_u(h\sigma)\chi_v(h\sigma)R_0^{\pm}(h;s,\sigma)S_0^{\pm}(h;s,\sigma)h(1+\sigma\kappa(s)h)\,\mathrm{d}\sigma.$$

Note that $F \in \mathscr{C}^{\infty}([0, \frac{L}{2\pi}] \times \mathbb{T}_L)$. Since $\theta'_0 = \hat{\tau}_0 > 0$, θ_0 is a smooth diffeomorphism form \mathbb{T}_L to \mathbb{T}_L , we perform the change of variable $x = \theta_0(s)$

$$\int_{\mathbb{T}_L} F(h;s) \, \mathrm{e}^{\mathrm{i}\frac{4\pi m}{L}\theta_0(s)} \, \mathrm{d}s = \int_{\mathbb{T}_L} (\theta_0^{-1})'(x) \, F(h;\theta_0^{-1}(x)) \, \, \mathrm{e}^{\mathrm{i}\frac{4\pi}{L}mx} \, \mathrm{d}x.$$

From the fact that the function $(h; x) \mapsto (\theta_0^{-1})'(x) F(h; \theta_0^{-1}(x)) \in \mathscr{C}^{\infty}([0, \frac{2\pi}{L}] \times \mathbb{T}_L)$ and the Riemann–Lebesgue lemma, we get

$$\int_{\mathbb{T}_L} (\theta_0^{-1})'(x) F(h; \theta_0^{-1}(x)) \, \mathrm{e}^{\mathrm{i} \frac{4\pi}{L} m x} \, \mathrm{d}x = \mathcal{O}(m^{-\infty}).$$

To prove Theorem 4.2, one just needs to establish that \underline{u}_m satisfies the first condition in Definition 4.1.

Proof of Theorem 4.2. Consider $(\underline{\lambda}_m, \underline{u}_m)_{m \ge 1}$ in Eq. (4.18), satisfying Lemma 4.12. We now define

$$\underline{\check{u}}_{m}(s,\xi) = \chi(\xi) \begin{cases} 0 & \text{if } \xi \leq 0\\ \left[\underline{u}_{m}\right]_{\mathbb{T}_{L} \times \{0\}}(s) + \xi \left[\varepsilon^{-1} \partial_{\xi} \underline{u}_{m}\right]_{\mathbb{T}_{L} \times \{0\}}(s) & \text{if } \xi > 0 \end{cases}$$

Using Lemma 4.12, we have $\|\underline{\check{u}}_m\|_{L^2(\mathbb{R}^2)} = \mathcal{O}(m^{-\infty})$ therefore $\underline{u}_m - \underline{\check{u}}_m \in D(P)$ and $(P - \underline{\lambda}_m)(\underline{u}_m - \underline{\check{u}}_m) = \mathcal{O}(m^{-\infty})$. We then replace \underline{u}_m by $\underline{u}_m = \frac{\underline{u}_m - \underline{\check{u}}_m}{\|\underline{u}_m - \underline{\check{u}}_m\|_{L^2(\mathbb{R}^2)}}$ which now makes $(\underline{\lambda}_m, \underline{u}_m)_{m \geq 1}$ a quasi-pair in the sense of Definition 4.1.

We finally show uniqueness of the quasi-resonances in the following sense:

Lemma 4.14. Let $(\underline{\lambda}_m, \underline{u}_m)_{m \geq 1}$ and $(\underline{\mu}_m, \underline{v}_m)_{m \geq 1}$ two quasi-pairs in the sense of *Definition 4.1* corresponding to the same integer m and having the same leading order phase $\theta_0 : s \mapsto \int_0^s \widehat{\tau}_0(t) dt$. Then we have the following estimate

$$\underline{\lambda}_m - \underline{\mu}_m = \mathcal{O}\left(m^{-\infty}\right)$$

Proof. Let \underline{R}_m , \underline{S}_m be the residuals $\underline{R}_m = P \underline{u}_m - \underline{\lambda}_m \underline{u}_m$, $\underline{S}_m = P \underline{v}_m - \underline{\mu}_m \underline{v}_m$. By definition, the residuals satisfy $\|\underline{R}_m\|_{L^2(\mathbb{R}^2)} = \mathcal{O}(m^{-\infty})$ and $\|\underline{S}_m\|_{L^2(\mathbb{R}^2)} = \mathcal{O}(m^{-\infty})$. Using the symmetry of the operator P, we get

$$\left(\underline{\lambda}_m - \underline{\overline{\mu}_m}\right) \int_{\mathbb{R}^2} \underline{u}_m \, \overline{\underline{v}_m} \, \mathrm{d}x = \int_{\mathbb{R}^2} \underline{u}_m \, \overline{\underline{S}_m} \, \mathrm{d}x - \int_{\mathbb{R}^2} \underline{R}_m \, \overline{\underline{v}_m} \, \mathrm{d}x = \mathcal{O}\left(m^{-\infty}\right)$$

From Remark 4.13 one can show that there exists $z_0 \in \mathbb{C}^*$ such that $\int_{\mathbb{R}^2} \underline{u}_m \, \overline{\underline{v}_m} \, \mathrm{d}x = z_0 + \mathcal{O}(m^{-1})$. Then $\underline{\lambda}_m - \underline{\mu}_m = \mathcal{O}(m^{-\infty})$ as $m \to +\infty$.

Corollary 4.15. The quasi-resonances $(\underline{\lambda}_m)_{m\geq 1}$ are real, and unique in the sense that if $(\underline{\lambda}_m)_{m\geq 1}$ and $(\underline{\mu}_m)_{m\geq 1}$ are two quasi-resonances, $\underline{\lambda}_m - \underline{\mu}_m = \mathcal{O}(m^{-\infty})$.

18

Proof. By applying Lemma 4.14 to $(\underline{\lambda}_m, \underline{u}_m)_{m\geq 1}$ and $(\underline{\lambda}_m, \underline{u}_m)_{m\geq 1}$, we get $\Im \underline{\lambda}_m = \mathcal{O}(m^{-\infty})$ which implies that $\Im \check{\lambda}_n = 0$ for all $n \in \mathbb{N}$. Then taking $(\underline{\lambda}_m, \underline{u}_m)_{m\geq 1}$ and $(\underline{\mu}_m, \underline{v}_m)_{m\geq 1}$ two quasi-pairs in the sense of Definition 4.1 having the same leading phase $\theta_0 : s \mapsto \int_0^s \widehat{\tau}_0(t) dt$, Lemma 4.14 and the fact that the quasi-resonances are real gives us $\underline{\lambda}_m - \underline{\mu}_m = \mathcal{O}(m^{-\infty})$.

Remark 4.16. With Corollary 4.15, given a quasi-pair $(\underline{\lambda}_m, \underline{u}_m)_{m\geq 1}$, we have a second quasi-orthogonal quasi-pair $(\underline{\lambda}_m, \overline{u}_m)_{m\geq 1}$ with the same quasi-resonance in the sense that, from (v) in Lemma 4.12,

$$\int_{\mathbb{R}^2} \underline{u}_m \ \overline{\underline{u}_m} \ \mathrm{d}x = \mathcal{O}\left(m^{-\infty}\right).$$

The quasi-resonances have an asymptotic multiplicity of 2, related to the chosen sign of the leading phase θ_0 (see Remark 4.8).

5. BACK TO THE SCATTERING PROBLEM

Now that we have constructed quasi-pairs characterizing surface plasmons, let's investigate their effect on scattering.

5.1. Instabilities and well-posedness. Let $\underline{\ell}_m \coloneqq \sqrt{\underline{\lambda}_m} \in \mathbb{C}^{\frac{1}{2}}$ for $m \geq 1$. The following theorem explains the instabilities observed in Section 3.1.

Theorem 5.1. If $\varepsilon_{\mathsf{c}}|_{\Gamma} < -1$, there exists m_0 such that for all $m \geq m_0$, we have $\underline{\ell}_m > 0$. Then along the quasi-resonances $(\underline{\ell}_m^2)_{m \geq m_0}$, constructed in Section 4, the stability constant $C(\underline{\ell}_m)$ of Problem (2.1) (see Eq. (A.2)) explodes super-algebraically, in the sense that for all $N \geq 0$, there exists $c_N > 0$ such that

$$C(\underline{\ell}_m) \ge c_N m^N, \qquad \forall m \ge 1.$$

Remark 5.2. From the definition of the function $N_{\varepsilon,\rho}$ and stability constant C(k), we have the estimate $N_{\varepsilon,\rho}(k) \leq C(k) \leq \sup_{u^{\text{in}} \in \mathcal{F}_{\rho}^{\text{in}} \setminus \{0\}} N_{\varepsilon,\rho}(k)$, where $\mathcal{F}_{\rho}^{\text{in}} = \{v \in H^1(\mathbb{D}(0,\rho)) \mid -(\Delta + k^2)v = 0\}.$

Proof. From Theorem 4.2, we have $-\operatorname{div}(\varepsilon^{-1} \nabla \underline{u}_m) - \underline{\ell}_m^2 \underline{u}_m = r_m$ with the remainder estimate $\|r_m\|_{L^2(\mathbb{R}^2)} = \mathcal{O}(m^{-\infty})$. Lemma A.1, with $u^{\text{in}} = 0$ and $f = r_m$, gives us

$$\|\underline{u}_m\|_{\mathrm{L}^2(\mathbb{R}^2)} \le C(\underline{\ell}_m) \|r_m\|_{\mathrm{L}^2(\mathbb{R}^2)}.$$

Since $\|\underline{u}_m\|_{L^2(\mathbb{R}^2)} = 1$ by definition and for all $N \ge 1$, there exists $\widetilde{c}_N > 0$ such that $\|r_m\|_{L^2(\mathbb{R}^2)} \le \widetilde{c}_N m^{-N}$ then $\widetilde{c}_N^{-1} m^N \le C(\underline{\ell}_m)$, for all $m \ge 1$.

We have now a systematic way to characterize asymptotically where are instabilities created by surface plasmons. In practice we compute the first order terms of the quasiresonances and provide intervals where those instabilities arise. In what follows we provide several numerical examples using Finite Element Method (FEM) to illustrate this result.

5.2. Numerical results. Using results from previous sections, we compute the first three terms of the quasi-resonances expansions, and compute related *plasmonic* intervals. From the expansion $\underline{\lambda}_m = \left(\frac{2\pi m}{L}\right)^2 \sum_{n=0}^2 \check{\lambda}_n \left(\frac{L}{2\pi m}\right)^n$, we deduce an expansion $\underline{\ell}_m \coloneqq \sqrt{\underline{\lambda}_m} = \frac{2\pi m}{L} \sum_{n=0}^2 \check{\ell}_n \left(\frac{L}{2\pi m}\right)^n$. In the case $\varepsilon_c < -1$, all the coefficients $\check{\ell}_n$ are real and from the expansion of $\underline{\ell}_m$, we denote the interval $I_m = [a_m, b_m]$ centered at $\underline{\ell}_m$ with

$$a_m = \frac{2\pi m}{L} \breve{\ell}_0 + \breve{\ell}_1 + \min\left(0, 2\breve{\ell}_2\right) \frac{L}{2\pi m} \quad \text{and} \quad b_m = \frac{2\pi m}{L} \breve{\ell}_0 + \breve{\ell}_1 + \max\left(0, 2\breve{\ell}_2\right) \frac{L}{2\pi m}.$$

We consider $\mathbb{D}(0, 1.5)$ as computational domain and various cavities $\Omega \subset \mathbb{D}(0, 1.5)$. We compute the function $k \mapsto N_{\varepsilon,1.5}$ from Problem (2.1), for various ε_c , with $u^{\text{in}}(x, y) = e^{iky}$ using FEM, and we check if instabilities can be captured in the intervals $(I_m)_{m\geq 1}$. The outgoing condition is imposed using a Dirichlet-to-Neumann map (DtN) [34] that considers 65 Fourier modes. All FEM computations are done with XLiFE++ [40], we use finite elements of order 7 on quadrangular structured meshes of order 3 constructed with GMSH [24], with embedded tubular neighborhood as defined in Section 4.2.1 (see Fig. 7 for some examples). Note that optimal FEM convergence is guaranteed as long as the mesh is locally symmetric along the interface Γ [9]. We consider three different cases summarized in Fig. 7:



FIGURE 7. Sketch representing the three considered configurations (a), (b), (c), for the numerical examples, and associated structured meshes: circular cavity (d), peanut shape cavity (e).

Case (a). Circular cavity of radius 1 with constant $\varepsilon_{\rm c}$ as represented in Fig. 7a with associated mesh in Fig. 7d. For the numerical examples we consider $\varepsilon_{\rm c} = -1.1$, a grid of 239 points for k, and 21736 degree of freedoms (dofs) for the FEM computations.

Case (b). Circular cavity of radius 1 with linearly varying permittivity $\varepsilon_{c}^{\varepsilon_{m},\varepsilon_{M}}:(x,y) \mapsto \frac{\varepsilon_{m}+\varepsilon_{M}}{2} + \frac{\varepsilon_{M}-\varepsilon_{m}}{2}x$ as represented in Fig. 7b with associated mesh in Fig. 7d. We consider $(\varepsilon_{m},\varepsilon_{M}) = (-1.2,-1.1)$, a grid of 216 points for k, and 21736 dofs for the FEM computations.

Case (c). Peanut cavity with constant ε_{c} as represented in Fig. 7c with associated mesh in Fig. 7e. The peanut boundary is parameterized by $r(\theta) = (1 - \frac{3}{10}\cos(2\theta))/L$, $\theta \in [0, 2\pi]$, with L such that Γ as length 2π . For the numerical examples we consider $\varepsilon_{c} = -1.1$, a grid of 237 points for k, and 27028 dofs for the FEM computations.

Figure 8 represents the function $k \mapsto N_{\varepsilon,1.5}(k)$ for the cases (a), (b), and (c) from left to right. The dotted orange lines correspond to FEM computations, the solid blue lines correspond to the analytic computation done in Section 3.1 (valid for case (a)). The purple zones correspond to the intervals $(I_m)_{m\geq 1}$ and the orange '×' correspond to local maxima. We observe:

- Figures 8a, 8b, and 8c present, as predicted by Theorem 5.1, instabilities manifested by sharp peaks where the scattering field is big.
- The observed peaks lie within the intervals I_m , for m sufficiently large (due to the asymptotic nature of the estimates). In other words, the intervals I_m are good estimates for spotting instabilities due to surface plasmons, for m large enough.
- While FEM captures instabilities, it fails to capture the peak's intensities: in particular in Fig. 8a FEM doesn't match analytic results, and in Fig. 8c FEM captures

sometimes local maxima so small it is not noticeable. In other words, while one can identify where surface plasmons arise using the intervals I_m , in practice FEM is unable to accurately capture them.

The provided asymptotic method allows us to identify regions where surface plasmons arise (responsible for large stability constants C(k)) for various metameterial cavities, and FEM computations confirm expected instabilities. The constructed quasiresonances provide guidance in order to avoid those instabilities. In practice, FEM is enable to accurately compute the emerging surface plasmons, extracting the asymptotic characterization (quasi-modes) could help the numerical method in that case.



FIGURE 8. Plot of the function $k \mapsto N_{\varepsilon,1.5}$ for the three cases (a), (b), and (c). The dotted orange lines correspond to FEM computations, the solid blue lines correspond to the analytic computation done in Section 3.1 (valid for case (a)). The purple zones correspond to the intervals I_m for $1 \leq m \leq 12$ and we have highlighted I_{12} . The '×' correspond to local maxima. We consider a uniform grid in k with geometric refinements in the intervals I_m centered at ℓ_m .

6. Conclusions

Similar to classical optical cavities, the scattering by metamaterial cavities can be significantly affected by localized waves. Those localized waves correspond here to emerging surface plasmons, which we have asymptotically characterized for various cavity properties (shape, permittivity, etc.). This study has been carried out without reducing to the quasi-static case, and the considered spectral parameter is the wavenumber by opposition to [25, 39, 1, 2]. Our asymptotic analysis revealed that the spectral nature of surface plasmons depends on whether $\varepsilon_{c} < -1$, or not: they are associated to negative eigenvalues when $-1 < \varepsilon_{c} < 0$ which can't be excited in practice, while they are associated to complex resonances when $\varepsilon_{\rm c} > -1$. Those resonances affect the stability constant and are responsible for numerical instabilities (large scattered field). FEM computations confirm that the asymptotic analysis provides good estimates where surface plasmons emerge, however FEM fails to accurately capture them. Similar to the singular complement method [20], one could consider extracting the asymptotic plasmonic behavior from the problem. Our asymptotic characterization of emerging surface plasmons can be applied to dispersive materials, namely while considering permittivity models depending on the wave number k, such as Drude's or Lorentz' model. In that context, our analysis confirms that surface plasmons can be excited only for frequencies lower than the surface plasmon frequency [31]. Extensions of the approach to polygonal metamaterial cavities and dispersive materials will be considered. In the quasi-static case, the spectral analysis for this case reveals hypersingular plasmonic behaviors and has been well investigated [26, 13]. The proposed approach is valid for arbitrary negative permittitivity ε_{c} , one could also consider arbitrary negative permeability μ_{c} and work with the double-negative PDE $-\operatorname{div}(\varepsilon^{-1}\nabla u) - \mu k^{2} u = 0$

(e.g. [11, 23, 3]). In that case the associated spectral problem becomes difficult because the operator is no longer self-adjoint. All the derivation has been provided for two-dimensional problems, one could consider three dimensional cavities. We have established that the existence of quasi-pairs implies the explosion of the stability constant when $\varepsilon_{\rm c} < -1$. Ongoing work focuses on proving that those quasi-pairs imply the existence of resonances close to the real axis via the *Black Box Scattering* [22].

APPENDIX A. Well-posedness of the scattering problem

Let $\mathbb{D}(0,\rho)$ be a disk a radius ρ such that $\Omega \in \mathbb{D}(0,\rho)$ and $f \in L^2(\mathbb{D}(0,\rho))$. Following [8], we use a Dirichlet-to-Neumann map, denoted \mathcal{S} , to rewrite a generalized version of Problem Eq. (2.1) in $\mathbb{D}(0,\rho)$ (with source f): Find $u^{sc} \in \mathrm{H}^1(\mathbb{D}(0,\rho))$ such that $u = u^{\text{in}} + u^{\text{sc}}$ and

$$\begin{cases} -\operatorname{div}\left(\varepsilon^{-1}\nabla u\right) - k^{2}u = f & \text{in } \mathbb{D}(0,\rho) \\ [u]_{\Gamma} = 0 & \text{and} & \left[\varepsilon^{-1}\partial_{n}u\right]_{\Gamma} = 0 & \text{across } \Gamma \\ \partial_{r}u - \mathcal{S}u = \partial_{r}u^{\text{in}} - \mathcal{S}u^{\text{in}} =: g^{\text{in}} & \text{across } \partial \mathbb{D}(0,\rho) \end{cases}$$
(A.1)

Problem Eq. (A.1) with $f \equiv 0$ is equivalent to Problem Eq. (2.1). The above general version will be useful in Section 4.

Lemma A.1. Problem (A.1) is well-posed if and only if $\varepsilon_{c}|_{\Gamma} \neq -1$. Moreover there exists a stability constant C(k) > 0 such that

$$\|u^{\mathsf{sc}}\|_{L^{2}(\mathbb{D}(0,\rho))} \leq C(k) \left(\|u^{\mathsf{in}}\|_{L^{2}(\mathbb{D}(0,\rho))} + \|f\|_{L^{2}(\mathbb{D}(0,\rho))} \right).$$
(A.2)

Proof. Consider $\varepsilon_{c} < 0$ constant. It has already been established that Problems Eq. (2.1)-Eq. (A.1) is well-posed (in Hadamard's sense) if and only if $\varepsilon_{c} \neq -1$ [8, Section 2]. The proof relies on T-coercivity arguments [10, 8, 9]. Consider now $\varepsilon_{\mathsf{c}} \in \mathscr{C}^{\infty}(\Omega)$ non constant. Lemma 1 in [8] establishes that problem Eq. (A.1) admits at most one solution in $H^1_{loc}(\mathbb{R}^2)$. One simply needs to establish that the operator $u \mapsto -\operatorname{div}(\varepsilon^{-1} \nabla u) - k^2 u$ is Fredholm if and only if $\varepsilon_{\mathsf{c}}|_{\Gamma} \neq -1$ to conclude. Since $\partial \Omega$ is a smooth interface, it can always be seen as locally straight with $\varepsilon_{\rm c}$ locally constant, then Theorems 4.3 and 6.2 in [10] apply and provide the needed results. Moreover, well-posedness gives us that there exists $\tilde{C}(k) > 0$ such that

$$\|u\|_{\mathrm{H}^{1}(\mathbb{D}(0,\rho))} \leq \widetilde{C}(k) \left(\left\|g^{\mathsf{in}}\right\|_{L^{2}(\mathbb{D}(0,\rho))} + \|f\|_{\mathrm{L}^{2}(\mathbb{D}(0,\rho))} \right),$$

which leads to

$$\|u^{\mathsf{sc}}\|_{L^{2}(\mathbb{D}(0,\rho))} \leq C(k) \left(\left\| u^{\mathsf{in}} \right\|_{L^{2}(\mathbb{D}(0,\rho))} + \|f\|_{L^{2}(\mathbb{D}(0,\rho))} \right)$$
(A.3)
of the Dirichlet-to-Neumann map and Poincar's inequality.

using properties of the Dirichlet-to-Neumann map and Poincar's inequality.

Remark A.2. Equation (A.3) can also be rewritten as $N_{\varepsilon,\rho} = \frac{\|u^{s\varepsilon}\|_{L^2_{\rho}}}{\|u^{in}\|_{L^2_{\rho}}} \leq C(k).$ In Section 3.1 we compute an approximation of $N_{\varepsilon,\rho}$. Computed results are then directly related to the well-posedness and the stability of the problem.

Appendix B. Proofs and additional results for the asymptotic EXPANSIONS

B.1. Proof of Lemma 4.6.

Proof. We solve Equation (4.15) as ordinary differential equations with $s \in \mathbb{T}_L$ as a parameter. The conditions $\varphi_0^{\pm}(s, \cdot) \in \mathscr{S}(\mathbb{R}_{\pm})$ give the following restrictions $\theta_0'(s)^2 + \varphi_0'(s)^2 + \varphi_0'(s)^2$ $\eta_0(s)^2 \check{\lambda}_0 \in \mathbb{C} \setminus \mathbb{R}_-$ and $\theta'_0(s)^2 - \check{\lambda}_0 \in \mathbb{C} \setminus \mathbb{R}_-$. If one of the above restrictions is false, then there are no solutions $\varphi^{\pm}(s,\cdot)$ in $\mathscr{S}(\mathbb{R}_{\pm})$. Under those restrictions, there exists $\alpha(s), \beta(s) \in \mathbb{R}$ such that $\alpha(s)\beta(s) \neq 0$ and

$$\varphi_0^-(s,\sigma) = \alpha(s) \exp\left(\sigma\sqrt{\theta_0'(s)^2 + \eta_0(s)^2\breve{\lambda}_0}\right),$$
$$\varphi_0^+(s,\sigma) = \beta(s) \exp\left(-\sigma\sqrt{\theta_0'(s)^2 - \breve{\lambda}_0}\right),$$

where the square roots are chosen to be in $\mathbb{C}^{\frac{1}{2}}$. The first transmission condition $\varphi_0^-(s,0) = \varphi_0^+(s,0)$ implies that $\alpha(s) = \beta(s) \neq 0$. Then the second transmission condition $-\eta_0(s)^{-2} \partial_\sigma \varphi_0^-(s,0) = \partial_\sigma \varphi_0^+(s,0)$ gives us

$$-\eta_0(s)^{-2}\sqrt{\theta_0'(s)^2 + \eta_0(s)^2}\check{\lambda}_0 = -\sqrt{\theta_0'(s)^2 - \check{\lambda}_0},$$

leading to the eikonal equation

$$\theta_0'(s)^2 = \frac{\check{\lambda}_0}{1 - \eta_0(s)^{-2}}.$$
 (B.1)

The right-hand side needs to be positive, leading to

$$\theta_0(s) = \int_0^s \sqrt{\breve{\lambda}_0 (1 - \eta_0(t)^{-2})^{-1}} \,\mathrm{d}t$$

and from the condition $\exp(ih^{-1}\theta_0) \in \mathscr{C}^{\infty}(\mathbb{T}_L)$, we deduce that $\exp(ih^{-1}\theta_0(L)) =$ $\exp(ih^{-1}\theta_0(0))$ which implies that there exists $m \in \mathbb{N}$ such that

$$2\pi m = h^{-1} \left(\theta_0(L) - \theta_0(0)\right) = h^{-1} \int_0^L \sqrt{\breve{\lambda}_0 \left(1 - \eta_0(s)^{-2}\right)^{-1}} \, \mathrm{d}s.$$

By choosing $h = \frac{L}{2\pi m}$ for $m \in \mathbb{N}^*$, we get

$$\left\langle \sqrt{\check{\lambda}_0 \left(1 - \eta_0^{-2}\right)^{-1}} \right\rangle = \left\langle \sqrt{\check{\lambda}_0 \varsigma \left|1 - \eta_0^{-2}\right|^{-1}} \right\rangle = 1$$

which gives $\check{\lambda}_0 = \varsigma \langle \tau_0 \rangle^{-2}$. Then with the relation $\tau_0^2 = \varsigma (1 - \eta_0^{-2})^{-1}$ we obtain that

$$\sqrt{\theta_0'(s)^2 + \eta_0(s)^2} \check{\lambda}_0 = \hat{\tau}_0(s) \eta_0(s) > 0 \quad \text{and} \quad \sqrt{\theta_0'(s)^2 - \check{\lambda}_0} = \hat{\tau}_0(s) \eta_0(s)^{-1} > 0,$$

ch concludes the proof.

which concludes the proof.

B.2. Proof of Lemma 4.9.

Proof. For $(s, \sigma) \in \mathbb{T}_L \times \mathbb{R}_{\pm}$, we define $\mathbf{e}^{\pm}(s, \sigma) = \exp\left(-|\sigma| \ \widehat{\tau}_0(s) \eta_0(s)^{\pm 1}\right)$. We proceed by induction on n. For n = 0, Lemma 4.6 gives $(\varphi_0^{\pm}, \theta_0, \breve{\lambda}_0)$ the solution of (\mathcal{P}_0) defined in Eq. (4.15). Let $n \geq 1$, from the definition of S_{n-1}^{\pm} in Eq. (4.17), there exists $Q_{n-1}^{\pm} \in \mathscr{C}^{\infty}(\mathbb{T}_L, \mathbb{P})$ such that $S_{n-1}^{\pm} = Q_{n-1}^{\pm} \mathbf{e}^{\pm}$ and using Lemma A.1 in [6], there exists $\widetilde{P}_n^{\pm} \in \mathscr{C}^{\infty}(\mathbb{T}_L, \mathbb{P})$ such that $\widetilde{\varphi}_n^{\pm} = \sigma \widetilde{P}_n^{\pm} \mathbf{e}^{\pm}$, $\partial_{\sigma}^2 \widetilde{\varphi}_n^- - \widehat{\tau}_0^2 \eta_0^2 \widetilde{\varphi}_n^- = \eta_0^2 S_{n-1}^-$, and

$$\partial_{\sigma}^{2} \widetilde{\varphi}_{n}^{+} - \widehat{\tau}_{0}^{2} \eta_{0}^{-2} \widetilde{\varphi}_{0}^{+} = -S_{n-1}^{+}. \text{ For } (s,\sigma) \in \mathbb{T}_{L} \times \mathbb{R}_{\pm}, \text{ we obtain}$$

$$\varphi_{n}^{-}(s,\sigma) = \alpha(s)\sigma \left(\frac{\eta_{0}(s) \breve{\lambda}_{n}}{2\widehat{\tau}_{0}(s)} + \frac{\theta_{n}'(s)}{\eta_{0}(s)} + \frac{\widetilde{P}_{n}^{-}(s,\sigma)}{\alpha(s)}\right) \mathbf{e}^{-}(s,\sigma),$$

$$\varphi_{n}^{+}(s,\sigma) = \alpha(s)\sigma \left(\frac{\eta_{0}(s) \breve{\lambda}_{n}}{2\widehat{\tau}_{0}(s)} - \eta_{0}(s) \theta_{n}'(s) + \frac{\widetilde{P}_{n}^{+}(s,\sigma)}{\alpha(s)}\right) \mathbf{e}^{+}(s,\sigma).$$

The first transmission condition $\varphi_n^-(\cdot, 0) = \varphi_n^+(\cdot, 0)$ is satisfied because $\varphi_n^{\pm}(\cdot, 0) = 0$. Using the second transmission condition $-\eta_0^{-2} \partial_\sigma \varphi_n^-(\cdot, 0) = \partial_\sigma \varphi_n^+(\cdot, 0)$ and the condition $\exp(i h^{n-1} \theta_n) \in \mathscr{C}^{\infty}(\mathbb{T}_L)$, we can choose

$$\theta_n(s) = \int_0^s \frac{\breve{\lambda}_n}{2\widehat{\tau}_0(t)(1-\eta_0(t)^{-2})} + \frac{\eta_0(t)\widetilde{P}_n^-(t,0) + \eta_0(t)^3\widetilde{P}_n^+(t,0)}{\alpha(t)(\eta_0(t)^4 - 1)} \,\mathrm{d}t \tag{B.2}$$

and with the relation $\tau_0(t)^2 (1 - \eta_0^{-2}) = \varsigma$ this give us

$$\breve{\lambda}_n = -\frac{2\varsigma}{\langle \tau_0 \rangle^2} \left\langle \frac{\eta_0 \widetilde{P}_n^-(\cdot, 0) + \eta_0^3 \widetilde{P}_n^+(\cdot, 0)}{\alpha \left(\eta_0^4 - 1\right)} \right\rangle.$$
(B.3)

Setting $P_n^{\pm}(s,\sigma) = \alpha(s)\sigma\left(\frac{\eta_0(s)\check{\lambda}_n}{2\hat{\tau}_0(s)} \mp \eta_0(s)^{\pm 1}\theta'_n(s) + \frac{\tilde{P}_n^{\pm}(s,\sigma)}{\alpha(s)}\right)$ finishes the proof. \Box

B.3. Additional results for Schwartz functions.

Lemma B.1. Consider $F : (h; s, \sigma) \mapsto F(h; s, \sigma)$ in $\mathscr{C}^{\infty}([0, \frac{L}{2\pi}] \times \mathbb{T}_L, \mathscr{S}(\mathbb{R}_{\pm})), \rho > 0$, and the intervals $I_{-}(h) = (-\infty, -\frac{\rho}{h})$ and $I_{+}(h) = (\frac{\rho}{h}, +\infty)$. Then $\int_{\mathbb{T}_L} \int_{I_{\pm}(h)} |F(h; s, \sigma)|^2 \, \mathrm{d}\sigma \, \mathrm{d}s = \mathcal{O}(h^{\infty}) \quad as \ h \to 0.$

Proof. Notice that, for any $N \ge 1$, there exists a constant $C_N > 0$ such that $|\sigma^N F(h; s, \sigma)| \le C_N$ for all $(h; s, \sigma) \in [0, \frac{L}{2\pi}] \times \mathbb{T}_L \times \mathbb{R}_{\pm}$. Hence,

$$\int_{\mathbb{T}_L} \int_{I_{\pm}(h)} |F(h; s, \sigma)|^2 \, \mathrm{d}\sigma \, \mathrm{d}s \le \frac{C_N L}{2N \, \rho^{2N-1}} \, h^{2N-1},$$

which finishes the proof.

B.4. Additional results used in Section 4.

Lemma B.2. For $s \in \mathbb{T}_L$,

$$\begin{aligned} \theta_1(s) &= \int_0^s \frac{\breve{\lambda}_1}{\breve{\lambda}_0} \widehat{\tau}_0(t) + \frac{(\eta_0(t)^2 - 1)\,\kappa(t)}{2\,\eta_0(t)} + \frac{\eta_1(t)}{2\,\eta_0(s)^2\,(\eta_0(t)^2 - 1)} \\ &+ \mathrm{i}\frac{(\eta_0(t)^4 + 3)\,\eta_0'(t)}{2\,\eta_0(t)\,(\eta_0(t)^4 - 1)} + \mathrm{i}\frac{\alpha'(t)}{\alpha(t)}\,\mathrm{d}t. \end{aligned}$$

Proof. This follows from Eq. (4.16) and Eq. (4.17).

Acknowledgments

The research was supported by the National Science Fundation Grant DMS-2009366. The authors would like to thank C. Tsogka and A. D. Kim for helpful feedback.

References

- H. AMMARI, Y. T. CHOW, AND H. LIU, Quantum ergodicity and localization of plasmon resonances, arXiv:2003.03696, (2020).
- [2] H. AMMARI, A. DABROWSKI, B. FITZPATRICK, AND P. MILLIEN, Perturbation of the scattering resonances of an open cavity by small particles. Part I: the transverse magnetic polarization case, Zeitschrift fr angewandte Mathematik und Physik, 71 (2020), p. 102.
- [3] H. AMMARI, B. FITZPATRICK, H. LEE, S. YU, AND H. ZHANG, Double-negative acoustic metamaterials, Quarterly of Applied Mathematics, 77 (2019), pp. 767–791.
- [4] H. AMMARI, P. MILLIEN, M. RUIZ, AND H. ZHANG, Mathematical analysis of plasmonic nanoparticles: The scalar case, Archive for Rational Mechanics and Analysis, 224 (2017), pp. 597– 658.
- [5] V. M. BABIC AND V. S. BULDYREV, Short-Wavelength Diffraction Theory: Asymptotic Methods, Springer Series on Wave Phenomena, Springer-Verlag, Berlin Heidelberg, 1972.
- [6] S. BALAC, M. DAUGE, AND Z. MOITIER, Asymptotics for 2d whispering gallery modes in optical micro-disks with radially varying index, 2020.
- [7] C. BERNARDI, M. DAUGE, AND Y. MADAY, Spectral methods for axisymmetric domains, vol. 3 of Series in Applied Mathematics (Paris), Gauthier-Villars, Éditions Scientifiques et Médicales Elsevier, Paris; North-Holland, Amsterdam, 1999.
- [8] A.-S. BONNET-BEN DHIA, C. CARVALHO, L. CHESNEL, AND P. CIARLET JR, On the use of perfectly matched layers at corners for scattering problems with sign-changing coefficients, Journal of Computational Physics, 322 (2016), pp. 224–247.
- [9] A.-S. BONNET-BEN DHIA, C. CARVALHO, AND P. CIARLET, Mesh requirements for the finite element approximation of problems with sign-changing coefficients, Numerische Mathematik, 138 (2018), pp. 801–838.
- [10] A.-S. BONNET-BEN DHIA, L. CHESNEL, AND P. CIARLET, *T-coercivity for scalar interface problems between dielectrics and metamaterials*, ESAIM: Mathematical Modelling and Numerical Analysis, 46 (2012), pp. 1363–1387.
- [11] A.-S. BONNET-BEN DHIA, L. CHESNEL, AND P. CIARLET JR, *T-Coercivity for the Maxwell Problem with Sign-Changing Coefficients*, Communications in Partial Differential Equations, 39 (2014), pp. 1007–1031.
- [12] A.-S. BONNET-BEN DHIA, L. CHESNEL, AND X. CLAEYS, Radiation condition for a non-smooth interface between a dielectric and a metamaterial, Mathematical Models and Methods in Applied Sciences, 23 (2013), pp. 1629–1662.
- [13] A.-S. BONNET-BEN DHIA, C. HAZARD, AND F. MONTEGHETTI, Complex-scaling method for the plasmonic resonances of planar subwavelength particles with corners. working paper or preprint, 2020.
- [14] E. BONNETIER, C. DAPOGNY, F. TRIKI, AND H. ZHANG, The plasmonic resonances of a bowtie antenna, arXiv preprint arXiv:1803.02614, (2018).
- [15] F. CAKONI AND D. COLTON, A Qualitative Approach to Inverse Scattering Theory, vol. 188 of Applied Mathematical Sciences, Springer, New York, 2014.
- [16] C. CARVALHO, L. CHESNEL, AND P. CIARLET JR, Eigenvalue problems with sign-changing coefficients, Comptes Rendus Mathematique, 355 (2017), pp. 671–675.
- [17] C. CARVALHO AND Z. MOITIER, Black box scattering for problems with sign-changing coefficients, in preparation, (2020).
- [18] L. CHESNEL, X. CLAEYS, AND S. A. NAZAROV, Spectrum of a diffusion operator with coefficient changing sign over a small inclusion, Zeitschrift f
 ür angewandte Mathematik und Physik, 66 (2015), pp. 2173–2196.
- [19] J. CHO, I. KIM, S. RIM, G.-S. YIM, AND C.-M. KIM, Outer resonances and effective potential analogy in two-dimensional dielectric cavities, Physics Letters A, 374 (2010), pp. 1893–1899.
- [20] P. CIARLET JR AND J. HE, The singular complement method for 2d scalar problems, Comptes Rendus Mathematique, 336 (2003), pp. 353–358.
- [21] D. COLTON AND R. KRESS, *Inverse acoustic and electromagnetic scattering theory*, vol. 93 of Applied Mathematical Sciences, Springer, New York, third ed., 2013.
- [22] S. DYATLOV AND M. ZWORSKI, *Mathematical Theory of Scattering Resonances*, American Mathematical Society, 2019.

- [23] B. FITZPATRICK, Mathematical Analysis of Minnaert Resonances for Acoustic Metamaterials, PhD thesis, ETH Zurich, 2018.
- [24] C. GEUZAINE AND J.-F. REMACLE, Gmsh: A 3-d finite element mesh generator with built-in pre- and post-processing facilities, International Journal for Numerical Methods in Engineering, 79 (2009), pp. 1309–1331.
- [25] D. GRIESER, The plasmonic eigenvalue problem, Reviews in Mathematical Physics, 26 (2014), p. 1450005.
- [26] C. HAZARD AND S. PAOLANTONI, Spectral analysis of polygonal cavities containing a negativeindex material, to appear in Annales Henri Lebesgue, (2021).
- [27] J. HEEBNER, R. GROVER, AND T. IBRAHIM, *Optical Microresonators: Theory, Fabrication, and Applications*, Springer Series in Optical Sciences, Springer-Verlag, New York, 2008.
- [28] J. HELSING AND A. KARLSSON, On a helmholtz transmission problem in planar domains with corners, Journal of Computational Physics, 371 (2018), pp. 315–332.
- [29] L. HÖRMANDER, The analysis of linear partial differential operators. I, vol. 256 of Grundlehren der Mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences], Springer-Verlag, Berlin, second ed., 1990.
- [30] P. KRAVANJA AND M. VAN BAREL, Computing the Zeros of Analytic Functions, Springer Berlin Heidelberg, 2000.
- [31] S. A. MAIER, Plasmonics: fundamentals and applications, Springer Science & Business Media, 2007.
- [32] A. MOIOLA AND E. A. SPENCE, Acoustic transmission problems: Wavenumber-explicit bounds and resonance-free regions, Mathematical Models and Methods in Applied Sciences, 29 (2019), pp. 317–354.
- [33] P. MOON AND D. E. SPENCER, Field Theory Handbook: Including Coordinate Systems, Differential Equations and Their Solutions, Springer-Verlag, Berlin Heidelberg, 2 ed., 1988.
- [34] A. A. OBERAI, M. MALHOTRA, AND P. M. PINSKY, On the implementation of the Dirichlet-to-Neumann radiation condition for iterative solution of the Helmholtz equation, Applied Numerical Mathematics, 27 (1998), pp. 443–464.
- [35] F. W. J. OLVER, D. W. LOZIER, R. F. BOISVERT, AND C. W. CLARK, eds., NIST handbook of mathematical functions, U.S. Department of Commerce, National Institute of Standards and Technology, Washington, DC; Cambridge University Press, Cambridge, 2010.
- [36] R. P., cxroots: A Python module to find all the roots of a complex analytic function within a given contour, 2018–.
- [37] G. C. RIGHINI, Y. DUMEIGE, P. FRON, M. FERRARI, G. NUNZI CONTI, D. RISTIC, AND S. SORIA, Whispering gallery mode microresonators: Fundamentals and applications, La Rivista del Nuovo Cimento, 34 (2011), pp. 435–488.
- [38] T. SANNOMIYA, C. HAFNER, AND J. VOROS, In situ sensing of single binding events by localized surface plasmon resonance, Nano letters, 8 (2008), pp. 3450–3455.
- [39] O. SCHNITZER, Geometric quantization of localized surface plasmons, IMA Journal of Applied Mathematics, 84 (2019), pp. 813–832.
- [40] XLIFE++, Librairie FEM-BEM C++, develope conjointement par les laboratoires IRMAR et POems. https://uma.ensta-paristech.fr/soft/XLiFE++/, 2010-.
- [41] M. ZWORSKI, Resonances in physics and geometry, Notices of the American Mathematical Society, 46 (1999), pp. 319–328.

DEPARTMENT OF APPLIED MATHEMATICS, UNIVERSITY OF CALIFORNIA, MERCED, 5200 NORTH LAKE ROAD, MERCED, CA 95343, USA

E-mail address: ccarvalho3@ucmerced.edu

DEPARTMENT OF APPLIED MATHEMATICS, UNIVERSITY OF CALIFORNIA, MERCED, 5200 NORTH LAKE ROAD, MERCED, CA 95343, USA

E-mail address: zmoitier@ucmerced.edu